ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Bergolo, Marcelo; Cruces, Guillermo

Working Paper Work and Tax Evasion Incentive Effects of Social Insurance Programs: Evidence from an Employment-Based Benefit Extension

IZA Discussion Papers, No. 8198

Provided in Cooperation with:

IZA – Institute of Labor Economics

Suggested Citation: Bergolo, Marcelo; Cruces, Guillermo (2014) : Work and Tax Evasion Incentive Effects of Social Insurance Programs: Evidence from an Employment-Based Benefit Extension, IZA Discussion Papers, No. 8198, Institute for the Study of Labor (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/98979

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

IZA DP No. 8198

Work and Tax Evasion Incentive Effects of Social Insurance Programs: Evidence from an Employment-Based Benefit Extension

Marcelo Bergolo Guillermo Cruces

May 2014

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Work and Tax Evasion Incentive Effects of Social Insurance Programs: Evidence from an Employment-Based Benefit Extension

Marcelo Bergolo

IECON-UDELAR and CEDLAS-UNLP

Guillermo Cruces

CEDLAS-UNLP, CONICET and IZA

Discussion Paper No. 8198 May 2014

IZA

P.O. Box 7240 53072 Bonn Germany

Phone: +49-228-3894-0 Fax: +49-228-3894-180 E-mail: iza@iza.org

Any opinions expressed here are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but the institute itself takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent nonprofit organization supported by Deutsche Post Foundation. The center is associated with the University of Bonn and offers a stimulating research environment through its international network, workshops and conferences, data service, project support, research visits and doctoral program. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

IZA Discussion Paper No. 8198 May 2014

ABSTRACT

Work and Tax Evasion Incentive Effects of Social Insurance Programs: Evidence from an Employment-Based Benefit Extension^{*}

This article studies how social insurance programs shape individual's incentives to take up registered employment and to report earnings to the tax authorities. The analysis is based on a social insurance reform in Uruguay that extended healthcare coverage to the dependent children of registered private-sector workers. The identification strategy relies on a comparison between individuals with and without dependent children before and after the reform. The reform increased benefit-eligible registered employment by 1.6 percentage points (about 5 percent above the pre-reform level), mainly due to an increase in labor force participation rather than to movement from unregistered to registered employment. The shift was greater for parents with younger children and for cohabiting adults whose partners' jobs did not provide the couples' children with access to the benefit. Finally, the reform increased the incidence of underreporting of salaried earnings by about 4 percentage points (25 percent higher than the pre-reform level), mostly for workers employed at small firms. The increase in fiscal revenue from higher levels of registered employment was several orders of magnitude greater than the loss of revenue due to an increase in underreporting.

JEL Classification: J22, H26, O17

Keywords: labor supply, work incentives, social insurance, tax evasion

Corresponding author:

Guillermo Cruces CEDLAS - Centro de Estudios Distributivos, Laborales y Sociales Facultad de Ciencias Económicas Universidad Nacional de La Plata Calle 6 entre 47 y 48, 5to. piso, oficina 516 1900 La Plata Argentina E-mail: gcruces@cedlas.org

^{*} The authors would like to thank Daniel Olesker and his staff at the Health Ministry for their comments and for providing administrative data on health insurance coverage. We also wish to thank the editor, Wojciech Kopczuk, and two anonymous referees for their substantial comments. We are grateful to María Laura Alzúa, Arnab Basu, Jim Berry, Mariano Bosch, Raj Chetty, Gary Fields, Leonardo Gasparini, David Kaplan, Carolina Lopez-Erazo, Michael Lovenheim, Marco Manacorda, Carmen Pagés and Norbert Schady for comments on earlier drafts of this paper, as well as seminar participants at the Cornell Development Workshop (February 2012) and the IZA Workshop on Recent Advances in Labor Supply Modeling (Dublin, May 2012).

1 Introduction

A large body of research on labor and public economics has examined the effects of social insurance and social welfare programs on incentives in the labor market. Most of the empirical literature has focused on behavioral responses to social insurance benefits (Krueger and Meyer, 2002) and to in-work benefits (Blundell and MaCurdy, 1999; Moffitt, 2003; Eissa and Hoynes, 2006). In many studies, emphasis has been placed on the financial incentive effects of labor force participation. The effects of responses on additional margins such as registered and unregistered employment and on underreporting of earnings to evade payroll taxes has received less attention (Slemrod and Yitzhaki, 2002; Sandmo, 2012: Slemrod and Weber, 2012). Understanding what leads firms and their employees to register with the formal system and to declare their true income to the government are first order public finance issues for most economies around the world. For developing and middle-income countries, this is an important issue because of widespread tax noncompliance, poor enforcement of regulations, and low levels of social insurance coverage. Higher levels of registered employment and more reporting of income subject to payroll taxes would result in increased access to social insurance for workers and their families. Moreover, a reduction in tax non-compliance could also help remedy a variety of market distortions that limit economic growth and economic development (La Porta and Shleifer, 2008; Levy, 2008). This issue is also relevant in advanced economies with smaller but still sizeable underground sectors (Schneider and Enste, 2000; Schneider, 2005).

This paper contributes to these discussions by exploring how social insurance programs shape individuals' incentives to take up registered employment and to alter the reporting of earnings to the tax authorities. This question is addressed empirically by examining a large-scale policy change in Uruguay's social insurance system. In this system, most benefits are made available only to employees registered with the social insurance administration who pay contributions and payroll taxes. Until 2007, a health insurance benefit was in place according to which the social insurance administration covered the employee's insurance premium from a private healthcare provider. However, the plan did not cover the employee's family. The 2008 reform extended the healthcare coverage funded by the SIA to include all registered employees' dependent children under the age of 18. The reform also increased the payroll tax contribution deducted from employees' salaried earnings, although, as described in detail below, the market value of the extended benefit was substantially greater than the increase in the tax liability for most workers with children.

This health insurance benefit extension (HIBEX) reform modified several margins of adjustment for eligible individuals (i.e., adults with dependent children). The reform introduced incentives to move into benefit-eligible employment (i.e., 25 hours or more per week at a registered job) to take advantage of the subsidy for dependent children's insurance premiums. The new benefit created an additional incentive for those not working to enter benefit-eligible employment, for those working in unregistered employment to formalize, and for registered workers below the 25-hour cut-off to increase their declared hours of work per week to at least 25.

The identification strategy relies on the fact that childless adults were not eligible for

the benefit extension. The reform's impact can thus be estimated using a difference-indifference approach, comparing the behavioral responses of adults with children to that of childless individuals before and after the reform. The analysis is based on a pool of repeated cross-sections of a nationally representative Uruguavan household survey that covers three years prior to and after the reform. The empirical analysis suggests that the benefit extension had a significant effect on labor market outcomes for eligible individuals. Specifically, the number of individuals in benefit-eligible registered employment increased by 1.6 percentage points with respect to the comparison group, about 5 percent higher than the pre-reform level. The evidence suggests that this aggregate effect was mostly the result of a reduction in the number of those not working rather than a consequence of increased work hours among registered employees below the threshold, or a movement of unregistered workers to registered employment. The effects were substantially larger for single parents, parents with several children, and parents with younger children, for whom the reform represented a stronger financial incentive. For single parents, a sizeable fraction of the effect can be attributed to movements from unregistered to registered employment. The response from individuals in couples depended on the benefit eligibility of their partners, with larger effects on individuals with a partner whose employment health insurance arrangements did not already offer coverage to the couple's children.

In addition to these responses in labor supply and employment registration, the reform could also have affected an additional margin of non-compliance, i.e, the underreporting of earnings to tax authorities to evade payroll tax and social insurance contributions. The HIBEX may have affected this form of non-compliance with the tax authorities because of the higher health insurance contribution rate (from three to six percent) and because those entering registered employment had to face the payroll taxes associated with the whole bundle of social insurance benefits (more than 30 percent of taxable earnings). The analysis below exploits a unique source of information on underreporting: a household survey in Uruguay asked registered employees whether their salaried earnings were being underreported to the tax authorities. Based on the plausibly exogenous variation introduced by the reform, the results presented below suggest that the extended benefit increased the incidence of underreporting of salaried earnings for individuals with children by about 4 percentage points, almost 25 percent higher than the pre-reform level. Moreover, this effect was confined to workers employed at small firms.

This paper contributes to the recent literature on labor supply, social insurance programs and tax compliance. First, it provides an analysis of individuals' reactions to social insurance policies in the labor market beyond the labor supply margin typically studied in the empirical literature (Ben-Shalom, Moffitt and Scholz, 2011). The analysis below also covers the effect of social insurance benefits on the registered/unregistered employment margin of adjustment. This is related to recent studies on the effects of work-based benefits such as the Earned Income Tax Credit (EITC) on formal employment (Potter Gunter, 2013). Because the EITC subsidizes only declared work, it induces individuals with low incomes to shift more hours from informal (or unregistered) to registered employment. This paper makes an analogous argument: since the additional benefit is only available through registered employment, the reform introduces an additional incentive to take up jobs that comply with tax regulations.

Second, this paper provides evidence to show that individuals also respond to the policy change through an additional margin of non-compliance, mainly the underreporting of salaried earnings. While the literature on tax evasion has typically concentrated on underreporting of self-employment income, in Uruguay (and in many other developing and middle-income countries) only employers report workers' salaried earnings, and they pay the full amount of taxes due.¹This analysis adds to the limited evidence currently available on underreporting and the incentives introduced by social insurance programs. The results presented here complement previous findings in at least two ways. On the one hand, rather than rely on indirect evidence such as the comparison between household income and expenditures (Tonin, 2011) or between income distributions from two separate sources (Kumler, Verhoogen and Frias, 2012), the findings are based on direct evidence of underreporting as stated by salaried workers. Moreover, the analysis also breaks down the impact of the reform on underreporting by firm size. This confirms the results of recent theoretical models (Kleven, Kreiner and Saez, 2009) and reinforces empirical findings found for other countries, including the United States (Chetty, Friedman and Saez, 2013). The direct evidence of adjustments to the policy reform in terms of off-the-book employment and underreporting of salaried earnings extends the range of evidence on tax evasion (Slemrod and Weber, 2012) and contributes to the literature that attempts to disentangle tax evasion from real behavioral responses to policies (Chetty, 2009; Saez, Slemrod and Giertz, 2012).

Finally, this paper provides one of the first estimations of the labor market response to social insurance benefits in Latin America. Bosch and Manacorda (2012) survey a relatively large body of empirical literature on the impact of non-contributive programs (such as conditional cash-transfers) on labor supply. They highlight the scarcity of evidence for the region on the effects of social insurance programs linked to registered employment. The findings presented here are in line with the theoretical literature on the effects of benefits tied to registered employment on individual choices between formal and informal employment (Levy, 2008; Galiani and Weinschelbaum, 2012).

This paper is organized as follows. Section 2 describes the institutional background of Uruguay's social insurance system and the 2008 healthcare insurance benefit extension. Section 3 details the work and tax evasion incentive effects introduced by the reform. Section 4 describes the data and the identification strategy to estimate the effects of the reform. Section 5 presents the main results on labor supply, registered employment, and underreporting of salaried earnings. Section 6 discusses a series of robustness and specification checks. Section 7 presents an analysis of the fiscal incidence of the reform accounting for its behavioral effects. Section 8 concludes with a discussion of the implications of these results for the analysis of tax and benefit systems in developing and advanced economies.

¹The empirical literature on underreporting of salaried earnings is scarce in part due to a lack of data. Moreover, this form of non-compliance is not deemed significant in developed countries because third-party information reporting is effective in inducing compliance, as suggested by the tax law lliterature (e.g. Shaw et al., 2010) and by empirical studies (e.g., Kleven et al., 2011).

2 Background: Related literature and Uruguay's social insurance system

2.1 Related literature

Uruguay's health insurance benefit extension (HIBEX) is one of a number of social insurance benefits linked to registered employment. The analysis of its effects on the labor market is related to a series of specific studies on benefits, work incentives and several margins of labor supply. With respect to labor force participation, the empirical literature generally finds a reduction in employment from increased social insurance benefits. Most of these studies, however, are related to social insurance benefits for the unemployed (Krueger and Meyer, 2002) or for other persons outside the labor force, including the disabled (Bound and Burkhauser, 1999). The analysis below covers these types of labor supply responses and expands the range of outcomes to include the impact of social insurance benefits on registered (i.e., on-the-book) employment and on misreporting of salaried earnings to the tax authorities.

The benefit-employment linkage also relates this reform to the literature on in-work welfare benefits such as the United States' Earned Income Tax Credits and the Temporary Assistance for Needy Families (Eissa and Liebman, 1996; Meyer and Rosenbaum, 2001) and the United Kingdom's Working Families Tax Credits (Francesconi and van der Klauuw, 2007). Studies of these programs typically find positive and significant effects on the extensive margin of labor supply, but the evidence is less conclusive on the intensive margin of response. An exception is the study by Chetty, Friedman and Saez (2013), who explain positive responses to the EITC in the intensive margin by means of an identification strategy that analyzes differences in knowledge about the tax code across neighborhoods. As in the analysis presented in this paper, most of these studies rely on comparisons between individuals with children and childless adults who are not affected by the policies and, hence, used as comparison groups.

This paper is also connected to the literature on the impact of extensions of welfare systems in developing countries, particularly in Latin America (Bosch and Manacorda, 2012). While most of these studies cover non-contributive programs, such as conditional social cash-transfer programs, this study investigates an expansion of a contributive social insurance benefit and its implications for labor supply incentives.

Finally, this study forms part of a growing body of research performed at the intersection between the literature on tax evasion-shadow employment and the literature on labor supply-tax and benefits in developing and developed countries. This type of analysis is based on Sandmo's (1981) and Cowell's (1985) extensions of the Allingham and Sandmo (1972) and Yitzhaki (1974) canonical models of tax evasion. In the context of Latin American labor markets, Levy (2008) and Galiani and Weinschelbaum (2012) model informality (the registered-non registered employment margin) in terms of tax evasion decisions as a function of fines and probabilities of detection. In a recent study for the United States, Potter Gunter (2013) relies on a quasi-experimental approach based on benefit variation by state to study the effect of the EITC on participation in regular (registered) and informal (unregistered or off-the-book) work among individuals with low incomes. The results indicate that higher levels of benefits in some states induce an increase in registered work and a fall in unregistered hours, with no significant effect on the extensive margin or on total hours worked.

The main outcome of interest in Potter Gunter's study (2013) relates to registered and unregistered self-employment. The analysis below studies unregistered salaried employment. Its innovation lies in examining another dimension of tax evasion, the underreporting of salaried earnings. The literature on tax evasion has typically focused on underreporting of earnings from self-employment because the withholding and reporting requirements of wage and salaried earnings make underreporting difficult (Slemrod and Weber, 2012) and quantitatively small, particularly in the United States (Slemrod, 2007). Most theoretical models also assume no evasion on wage and salary income (Feldman and Slemrod, 2007). Paying a fraction (but not all) of earnings under the table is a potentially relevant margin of adjustment. Kleven, Kreiner and Saez (2009) develop a theoretical model where workers and firms may collude to underreport earnings to the tax authority. The model predicts that this type of collusion may be feasible at small firms, but harder to sustain on a larger scale. Tonin (2011) builds a theoretical model where firms and workers collude to hide part or all of the employee's salaried earnings to the fiscal authority "to avoid the payment of taxes and social security contributions." The results in this paper are consistent with these types of models. They are also directly related to recent attempts to empirically test their predictions by analyzing the impact of changes in taxes and benefits on misreporting of social insurance contributions. Tonin (2011) presents supporting evidence based on household income and consumption patterns in Hungary following a hike in the minimum wage in 2001. Consistent with the model, the hike induced higher compliance and reduced disposable income for skilled workers.

Underreporting of salaried earnings, however, might also be decided solely by the firm rather than pursuant to collusion, especially in developing countries where there are no third-party reporting requirements. This lack of information on reporting² implies that the underreporting of salaried earnings may be a quantitatively relevant issue (see Kleven et al., 2011 for a discussion of third-party information reporting). Kumler, Verhoogen and Frías (2012) study underreporting of earnings in Mexico by examining a pension reform that transitioned from a pay-as-you-go system to a system based on fully funded individual accounts. The reform gave employees an additional incentive to ensure accurate reports of earnings by firms and to monitor their employers' compliance with social insurance contributions. A comparison of income distributions from administrative data and household surveys reveals substantial underreporting of earnings by firms with lower levels of underreporting for larger firms. Moreover, a difference in differences analysis comparing younger cohorts (which benefit more from the reform) to older workers indicates a significant decline in underreporting after the reform.

This paper complements these studies of underreporting in at least two ways. Instead

 $^{^{2}}$ Strictly speaking, this would be a lack of first-party reporting. The tax authority only has employers' reports for most employees' earnings. The result is the same: the tax authority does not have two sources of information to compare.

of relying on comparisons between household income and expenditures or between income distributions from two separate sources, the analysis below relies on direct evidence of underreporting of earnings as stated by workers. Moreover, the analysis demonstrates the impact of the reform on underreporting of salaried earnings, breaking down the effect by firm size, as suggested in the recent literature (Chetty et al., 2013; Kleven et al., 2009).

2.2 Uruguay's social insurance system

Uruguay has one of the oldest and most developed social insurance systems in Latin America. The following is a review of only the aspects of that system that are most relevant to this paper; for a more in-depth description, see Ferreira-Coimbra and Forteza (2004). This system follows a European Bismark-type model, where access to most social insurance programs is linked to registered employment and financed through payroll taxes and contributions from registered employees. The *Banco de Previsión Social* (BPS) is the institution in charge of most of its administration (henceforth, the social insurance administration - SIA). It collects payroll taxes and contributions and provides social insurance benefits to most private-sector employees and public-sector workers. These benefits include retirement pension plans and a wide range of other programs, such as health insurance, disability benefits, maternity leave and family allowances (cash transfers paid for each of the worker's dependent children).

Private-sector employers in Uruguay are required to register their employees with the SIA. Both the employer and employee are required to make contributions to finance social insurance programs. Formally, the employer deducts the employee's contributions from her salary and pays both the employee's and the employer's contributions to the SIA. Table 1, Panel B, shows the breakdown of the SIA contributions required by law as a percentage of the taxable salary for private-sector employers and employees. Although the total consists of contributions to different programs, it is transferred in full by the employer to the SIA. In many senses, then, it operates like an overall payroll tax (the paper uses contribution and payroll tax interchangeably). This payroll tax amounted to about 31 percent of taxable wages in 2007, the year before the reform analyzed in this paper. The social insurance benefits are provided as an indivisible bundle, that is, workers or employers cannot opt out of certain programs and pocket the correspondent contribution, although there are minimum work requirements for some benefits as discussed below. Therefore registered employment - that is, employment compliant with social insurance regulations and payroll taxes - grants the worker access to the full bundle of benefits.

As in many middle-income countries, enforcement in Uruguay is relatively weak and there is widespread non-compliance with social insurance regulations and evasion of the payroll tax. This means that a substantial fraction of private-sector salaried employees lack access to employment-related social insurance benefits. Unregistered or off-the-book workers in Uruguay constituted about a quarter of total salaried employees in the decade of 2000 (Gasparini and Tornarolli, 2009) and about 28 percent of private-sector employees in 2007, as computed with microdata from the *Encuesta Continua de Hogares* described below. This phenomenon is sometimes referred to as labor informality. This paper uses a more precise terminology based on the situation of the worker with respect to the SIA. The discussion refers to an employee's registration status or, alternatively, to the employer and employee's compliance with payroll taxes and social insurance regulations.

2.3 The 2008 reform: The health insurance benefit extension (HIBEX)

Eligibility for the health insurance component of the social insurance system requires workers to be registered with the SIA and to work 25 or more hours per week.³ Qualifying employees have the right to select a health insurance provider in the private sector, which are structured like HMOs in the United States. The insurance premium is paid by the SIA. This health insurance package covers services for the worker that are similar to those offered in the private healthcare market, usually by the same institutions. These packages include inpatient hospital coverage, emergency room care, medical and surgical services, maternity care, among other services. By 2007, the health insurance benefit covered 43 percent of the labor force and approximately 66 percent of private- sector salaried employees (figures based on the *Encuesta Continua de Hogares*). That year, the SIA spent about UYU 7.4 billion (USD 471 million at the 2010 PPP adjusted exchange rate) on health insurance premiums, which represented 13 percent of its budget and 1.4 percent of the Uruguay's gross domestic product (Banco de Previsión Social, 2008). In addition, the health insurance benefit for employees represented about 16 percent of total healthcare expenditures in Uruguay in 2007 (Ministerio de Salud Pública, 2010a).

In 2007, Uruguay's government embarked on an ambitious overhaul of the healthcare system, with the stated objective of strengthening three areas: healthcare coverage; health management; and healthcare financing. The reform enacted by Uruguay's Parliament (Law 18,211-2007) established a series of new guidelines, regulations and institutions for the healthcare system in order to further equitable access.

Before the 2008 reform, the health insurance benefit was financed by contributions to the SIA amounting to 8 percent of taxable salaries; five percent represented the employer's contribution and 3 percent the employee's (Table 1, Panel B). The health insurance benefit, however, covered only registered employees. Individuals not in the labor force, unregistered employees, workers' spouses who were not themselves registered employees, and the dependent children of registered employees had to pay for their own health care, purchase private health insurance, or use the public healthcare system, which is subject to a means test.

The most important provision of the reform was the extension of SIA health insurance benefits to cover the beneficiary's dependent spouses and children.⁴ Because of fiscal

³Very low-income workers are also excluded from the health insurance benefit. The threshold is defined as those earning below 1.25 times the *Base de Prestaciones y Contribuciones* (BPC). Created in 2004 (Law 17,856), the BPC index replaced the minimum wage as the indexing basis for all government benefits. The baseline was the value of the minimum wage in force on December 20, 2004. It was UYU 1,636 in 2007, or about USD 103 per month at the 2010 PPP adjusted exchange rate.

⁴Borgia (2008) and Bergolo and Cruces (2013) provide more details about the healthcare system reform, including the benefit expansion discussed here. Cruces and Bergolo (2013) describe the series of recent reforms to Uruguay's social insurance system, including health insurance, family allowances and pensions.

constraints, this extension was implemented in stages. For the period covered in this paper, the expansion only reached the dependent children of registered employees.⁵ The reform was passed in December 2007 and took effect on January 1, 2008. Eligibility for the health insurance benefit extension required that the registered employee have at least one dependent child under the age of 18.⁶ The inclusion of these children under the extended benefit was universal, irrespective of pre-existing health conditions and previous health insurance arrangements. As with the previous individual-based benefits, the worker had to be registered with the SIA (compliant with payroll taxes and social insurance regulations), and work at least 25 hours per week or earn above a minimum threshold. While the reform covered all salaried workers, the HIBEX was most relevant for private-sector registered employees, since a large fraction of public-sector workers were already entitled to health insurance coverage for their dependents (Ministerio de Salud Pública, 2010b).

HIBEX beneficiaries could choose a health insurance provider in the private sector or rely on the public healthcare system for their children. Regardless of of the option chosen, the SIA reimbursed each child's premium to the corresponding institutions. Registered employees, however, opted overwhelmingly for the private sector, with only about 5 percent of all children enrolled in the extended benefit using the public system in 2008 (Banco de Previsión Social, 2009).

The HIBEX represented a large-scale policy change. It had a substantial impact on the health insurance coverage of its target population and substantially increased privately provided insurance for children under the age of 18. Figure 1 depicts the trend in the percentage of dependent children of private-sector employees aged 0-17 (i.e., eligible for HIBEX) who used private-sector healthcare providers for the period 2004-2010; from 2008 onwards, it depicts the trend in the proportion of those who accessed private-sector health care through the HIBEX. The figure also plots the same series for individuals aged 18 and older - i.e., those not covered by the benefit extension.⁷ In the period from 2004 to 2007, the evolution of private healthcare coverage for both groups was similar. Following the 2008 reform, private healthcare coverage for individuals aged 0-17 increased by 30 percentage points, and it continued to grow (albeit slightly) until 2010. This change was driven by the HIBEX reform, as indicated by the series depicting the percentage of children 0-17 who accessed private health care through the new program. In contrast, there was no significant change in the trend in private healthcare coverage for individuals aged 18 or older. By 2008, 450,000 children under 18 years old were insured through the HIBEX, or about 48 percent of all children of this age group in Uruguay (Banco de Previsión Social,

 $^{^{5}}$ The extended benefit also reached children under the age of 18 with retired parents eligible for a SIA old-age pension after the reform. It covered as well the dependent children of those retired before the reform but who belonged to the low-income group of the old-age pension system. By 2008, children in this group represented fewer than 1 percent of all children enrolled (Banco de Previsión Social, 2009). This small number of individuals is excluded from the analysis.

⁶For the purpose of the SIA benefit, a dependent child is defined as an offspring, a stepchild or a foster child of the registered employee under the age of 18. Disabled children above this age threshold also qualify for the HIBEX. The reform also established that dependent children enrolled in the HIBEX could retain coverage until the age of 21 if their parents reimbursed the value of the child's insurance premium to the SIA.

⁷Computed with microdata from *the Encuesta Continua de Hogares*, described below. Similar trends are found in administrative records from the Ministry for Public Health.

2009) out of a total population of about 3.3 million individuals (Uruguayan National Bureau of Statistics, projection for 2008).

To finance this extension of coverage, the legislation specified an increase in payroll contributions for most workers, with a crucial distinction between parents and childless workers. Employer contributions remained unchanged by the reform at 5 percent of taxable earnings as one of the reform's stated intentions was to leave firms' employment costs unaffected. The employee's contribution to the health insurance benefit, however, increased from 3 to 6 percent of taxable earnings for registered private-sector workers with dependent children under the age of 18, regardless of the number of qualifying children. The reform also increased the contribution for childless registered employees and for those whose children did not qualify (those over the age of 18) from 3 to 4.5 percent of their taxable earnings. The payroll tax remained unchanged at the 3 percent level for low-wage workers, defined as those with monthly earnings below 2.5 times the value of the BPC (UYU 1,775 per month in 2008, USD 112), irrespective of whether they had children who qualified for the HIBEX or not. The reform thus increased the overall payroll tax rate from 30.75 to 32.25 percent for most childless registered workers, and from 30.75 to 33.75 percent for most registered workers with young dependent children, with no change in contributions for very low-wage workers. Table 1, Panel B, summarizes the changes in the contributions and the total payroll taxes introduced by the HIBEX reform.

The reform implied a potential net benefit for most adults with eligible children. The market value of the extended benefit exceeded the increase in the tax liability for most parents in registered employment. For instance, the monthly health insurance premium for a child in the private market in 2007 was more than two times the 3 percent increase in the contribution of the average monthly taxable earnings of a registered private-sector worker.⁸ Moreover, the average number of eligible children per adult in the sample in 2007 was 1.92 (Table 1). However, this comparison must be qualified. For high earners with few children, the cost of the increase in the payroll tax may have ended up being more than the new benefit, although only a small number of workers at the top of the income distribution faced this potential net loss.⁹A further qualification stems from the fact that childless workers faced an increase in the cost of registered employment (a total of 1.5 percent of salaried earnings) without any additional benefit. This may have yielded a reduction in the level of employment for this group of workers. However, the available empirical evidence for Latin America indicates that even sweeping changes in labor taxes had negligible effects on employment (Gruber, 1997; Cruces, Galiani and Kidyba, 2010). An increase of 1.5 percentage points in the social insurance contributions of childless registered workers can, then, be expected to have had a negligible effect. In any case,

⁸The average monthly taxable earnings for a private-sector registered employee in 2007 was UYU 12,305 (USD 778 per month at the 2010 PPP adjusted exchange rate), and thus the 3 percentage point increase in the contribution rate represented UYU 369 (USD 23). In contrast, the health insurance premium for a child in the private market in 2007 was on average UYU 859 (USD 54) per month (Ministerio de Salud Pública, 2010a).

⁹In 2007, the reform implied a net loss in income only for those registered workers with one child and taxable earnings above UYU 28,633, and for those with two children and earnings above UYU 57,267. These values correspond to the 94th and 99th percentiles of the distribution of monthly labor earnings for registered workers in 2007.

after the reform, only about 8 percent of children benefiting from the HIBEX used the public healthcare sector rather than private health insurance providers (Figure 1). The very high take up of the HIBEX suggests that it represented a net benefit for a majority of those eligible. Thus, the empirical analysis in this paper is based on a comparison of labor market outcomes for adults with and without eligible children. Section 4.2 below addresses some of the caveats with this comparison, the identification strategy and some robustness tests to validate it.

3 Work and tax evasion incentives effects of the health insurance benefit extension

The health insurance benefit extension represents an increase in the benefits associated with registered employment for individuals who work 25 hours or more per week (the minimum hour cut-off for eligibility) and have children under the age of 18. This policy change yields a number of predictions on labor supply and tax evasion responses.¹⁰ Given these new incentives, the main expected effect of the reform would be an increase in the probability of moving into benefit-eligible employment (i.e., working 25 hours or more per week in registered employment) for parents of children in the 0-17 age range. There are three possible channels through which this change could operate. First, the reform introduced an incentive to work in registered employment with 25 or more hours per week for those not working. Second, those in unregistered employment faced a similar incentive: either to move into a benefit-eligible job or to formalize their existing employment arrangement (i.e., getting "on"-the-books with 25 declared hours at their current job). Third, the reform provided an incentive for those in registered employment working less than 25 hours per week to increase their declared hours of work to at least 25.

The main expected effect yields the prediction that (1) the number of individuals with children in registered employment working 25 hours or more per week should increase after the reform. The different channels through which this main effect operates imply some of the following additional predictions: (2) the number of those not working should decrease, (3) the number of individuals with unregistered employment should fall, (4) the number of individuals in registered employment (irrespective of hours worked) should increase, and more specifically (5) conditioned on registered employment, the number of those working 25 hours or more per week should increase.¹¹

While the discussion has thus far referred to overall effects, the benefits derived from the reform (and thus the behavioral reactions to the new incentives) may have varied in magnitude for different subgroups defined by the demographic characteristics of an individual's household. First, since the increase in the payroll tax was independent of the number of children, the reform implied a larger financial incentive for individuals with

¹⁰The structure of the discussion in this section was suggested by an anonymous referee.

¹¹These are all quantitatively relevant margins of adjustment. In 2007, before the reform, about 54 percent of individuals with children in the sample were not eligible for the extended benefit. Those not working represented 33 percent of all individuals with children, 16 percent were unregistered employees, and about 5 percent were registered employees working below the 25 weeklyhour threshold (see Table 1, Panel A).

more children and for those with younger children (since the benefit covers children up to the age of 18, implying a longer potential duration of the benefit).

Second, the rules of the benefit extension imply that the HIBEX may have induced a differential response for single parents compared to parents living with their partners for at least two reasons.¹² On the one hand, the rules of the reform implied a potential double taxation for couples, since if the two parents were registered employees, both had to pay the increase in the payroll tax (if both worked more than 25 hours a week) for the same level of benefit. Thus, individuals with partners in benefit-eligible employment did not face an additional incentive to move to benefit-eligible employment themselves after the reform, since the couple's children would be covered through their partners (in fact, they faced an incentive to reduce participation in benefit-eligible employment). In turn, individuals whose partners were not in benefit-eligible employment faced an additional incentive to take up (or remain in) benefit-eligible employment themselves to warrant coverage for the couple's children. On the other hand, the HIBEX represented an unambiguous incentive for single parents, in contrast to married individuals, to take up benefit-eligible employment, since they would not face any double taxation or have to take another individual's labor market outcomes into account when making their own decisions.

Finally, changes in tax rates can also affect the incentives for reporting earnings to the tax authorities. Given the increased costs associated with registered employment in the form of higher payroll taxes, the reform may also have affected the intensive margin of tax evasion by inducing more underreporting of salaried earnings. This additional incentive may affect both the firm (which faces higher payroll tax liabilities) and the worker (who faces lower after-tax wages). It may result in higher rates of underreporting on the part of the firm or pursuant to collusion of employers and employees. In any case, the literature reviewed above indicates that underreporting is more likely at small firms. An additional prediction on the effects of the reform is thus that (6) the number of registered employees with eligible children stating that their earnings were underreported should increase, especially among those working at small firms.

4 Data and empirical strategy

4.1 Sample and outcomes of interest

The empirical analysis in this paper is based on cross-sectional microdata from the *Encuesta Continua de Hogares* (ECH) survey carried out by Uruguay's National Institute of Statistics (*Instituto Nacional de Estadística*, INE) since 1990. The ECH is a nationally representative household survey conducted according to international standards. It combines elements of living standards and labor force surveys, and it is the main source for socioeconomic, labor and demographic indicators in Uruguay. The microdata and supporting documents, such as questionnaires and details on sample selection and stratification, are all publicly available.

¹²The benefit extension covered the children of legally married and cohabiting couples. The text sometimes refers to individuals in both types of arrangements as married for the sake of brevity.

The analysis in this paper relies on two waves of the survey corresponding to the first and second semester of the calendar year. These repeated cross-sections cover the period from the 2nd semester of 2004 to the 2nd semester of 2010, that is, three and a half years before the reform and three years after it. The sample for the analysis includes men and women aged 25 through 55 who are either heads of households or ' spouses of heads of households. The sample excludes individuals who reside in rural areas or in urban areas with less than 5000 inhabitants, since the ECH only extended its coverage to such areas in 2006.¹³ The analysis is carried out on a sample of private-sector salaried employees (excluding the self-employed), as well as the unemployed and individuals outside the labor force in the relevant age range. The sample does not include government employees because a large proportion were entitled to health insurance for their dependents before the reform. The sample also excludes retired and disabled individuals and those who were studying full time during the survey's referenceperiod. Taking all of these restrictions into account and pooling all the survey waves yields a sample of 97,552 individuals for the 2004-2010 period.

The ECH collects information on all the relevant outcomes discussed in the previous section: employment and labor force participation, hours of work, the SIA registration status of the employee, salaried earnings and whether there is underreporting of such earnings. The registration status of an employee is derived from a specific self-reported variable that indicates whether contributions (payroll taxes) are paid to the SIA for her current job.¹⁴ The ECH questionnaire also contains a unique feature for detecting the presence of underreporting of earnings for registered salaried workers. Since 2006, registered private-sector employees have been asked directly whether their salaried earnings are being underreported to the tax and social security authorities. The specific question is: "Are your contributions based on the total amount of your earnings from this job?" ("¿Aporta por la totalidad del salario en esa ocupación?"). The explicit aim of the question is to establish if tax evasion is taking place, as stated in the ECH enumerator's manual which specifies that this question attempts to "capture underreporting of salaried earnings for the computation of social security contributions".¹⁵ Finally, all of the estimates presented below are computed using the corresponding ECH sampling weights.¹⁶

 $^{^{13}}$ While this implies a loss of information, there is a trade-off between a more representative sample and a longer pre-reform period with a comparable sample. The latter option was chosen since more than 80 percent of the population of Uruguay resides in urban areas with more than 5000 inhabitants.

¹⁴The specific question in the ECH is: "Are you contributing to a retirement benefit through this job?" ("¿Aporta a una caja de jubilaciones por este trabajo?"). This is a standardized question in household surveys in Latin America; it is used to define registered or formal work in most of the recent literature for the region - see, for instance, Gasparini and Tornarolli (2009) and Galiani and Weinschelbaum (2012).

¹⁵This question refers specifically to underreporting, and it does not reflect a cap on contributions, which are voluntary for earnings above a relatively high threshold (monthly earnings above UYU 74,000, about USD 3,800). This is confirmed by the fact that individuals self-report as underreporting earnings along the whole income distribution. For instance, for the sample before the reform, 15.1 percent of individuals in the bottom quintile of the distribution of monthly income declared underreporting of salaried earnings, with 11.6, 10.9, 9.9 and 6.0 percent in the following quintiles.

¹⁶The main results do not vary when the regressions are estimated without the ECH sampling weights.

4.2 Identification assumptions, econometric strategy and outcomes of interest

The empirical work in this paper attempts to identify the causal effect of the introduction of the 2008 health reform on labor market outcomes and tax evasion and to provide evidence on the direction and magnitude of the incentive effects discussed in the previous section. The strategy used to evaluate the policy's effects is a difference-in-differences analysis (Angrist and Krueger, 1999). This framework compares the evolution of the outcomes of interest for a group of individuals exposed to a policy change (the treatment group) to the change in the same outcomes for individuals unaffected by the reform (the comparison group). The latter group is assumed to capture the counterfactual trend for the treatment group which would have been observed in the absence of the policy change.

The treatment group consists of individuals aged 25 to 55 with at least one dependent child under the age of 18, - that is, those potentially eligible for the HIBEX. The comparison group consists of childless individuals in the same age range who were not directly affected by the benefit extension.¹⁷ The empirical analysis therefore compares adults with at least one dependent child under the age of 18 to individuals in the same age group with no dependent children. The following is the basic difference-in-difference specification with controls on which most of the estimates in this paper are based:

$Y_{ijt} = \alpha + \delta AnyChildren_{ijt} + \beta AnyChildren_{ijt} * PostReform_t + X'_{ijt}\gamma + \theta_t + \varphi_j + \tau_{jt} + \epsilon_{ijt}$ (1)

where *i* indexes individuals, *j* state and *t* time. The variable Y_{ijt} is one of the outcomes of interest (described in more detail below); $AnyChildren_{ijt}$ is an indicator variable for individuals in the treatment group, coded as one if the individual has at least one child under the age of 18 and zero otherwise; $PostReform_t$ is a dummy equal to one in the post-policy period (from January 2008 onwards) and zero otherwise; and $AnyChildren_{ijt} * PostReform_t$ is the interaction term between the two variables, which captures the difference-in-difference treatment effect. The X_{it} matrix contains individualspecific variables to condition the differences in trends to observable characteristics. The individual controls are age (in 6 categories), gender, head of the household indicator, marital status, education level (in 6 categories), the number children in the household, the number of children under the age of 6, and the number of children and age and education interactions (both in 6 categories). The estimations also include controls for firm size and industry (5 and 9 categories, respectively) for the outcomes which are defined for the labor force. θ_t and φ_j are a full set of semester and state fixed effects (Uruguay is administratively divided into nineteen states or "departamentos"), which account for aggregate systematic shocks to local labor markets correlated with, but not caused by, the

¹⁷The empirical analysis below also relies on an alternative comparison between individuals with and without children within the group of low-earning registered employees (those earning below 2.5 BPC). These individuals benefited from the extended coverage if they had children, but did not face the increase in SIA contributions (from 3 to 6 percent for those with children and from 3 to 4.5 percent for the childless). While this constitutes a potentially viable comparison group, the resulting sample size is small and the results are underpowered. These results are presented as part of the robustness tests in Table 7 below.

HIBEX. Finally, τ_{jt} are state-by-semester effects, introduced to control for state-specific shocks over the period that might be independent of national economic conditions.

The analysis is based on six outcomes of interest directly related to the expected effects discussed in Section 3. The first is benefit-eligible employment, which is an indicator coded as one for salaried employees who report 25 hours or more per week at a registered job and zero otherwise. This minimum hour cut-off is determined by the eligibility requirement for health insurance coverage. The second outcome of interest is registered employment, which is an indicator variable coded as one for registered salaried employees and zero otherwise (irrespective of the number of hours worked). The third outcome is unregistered employment, an indicator coded as one for off-the-book salaried employees and zero otherwise. The fourth variable of interest is benefit-eligible employment defined only for registered employees, which is coded as one for registered employment at or above 25 hours per week and zero for those in registered employment working 24 hours or less. A fifth outcome of interest is an indicator variable equal to one for those not working and zero for those working (either as registered or unregistered employees). These variables are defined for all individuals in the sample for the whole 2004-2010 period, with the exception of conditional benefit-eligible employment which is defined only for the subsample of registered employees (56,707 observations). Finally, a sixth outcome of interest, underreporting of salaried earnings, is captured by a measure coded as one if the individual reports her salaried earnings as underreported to the SIA and zero otherwise. The analysis is based on the subsample of registered employees for the years 2006-2010 (50,669 observations).

The estimations rely on linear probability models (OLS) for the binary dependent variables and all the regression results report Huber-White robust standard errors. The main results are based on the 2004-2010 period excluding both semesters of 2008, since the policy change was effectively implemented throughout that year.¹⁸

The difference-in-differences identification strategy in this setting requires two main assumptions. First, in the absence of the reform, the underlying trends in labor market outcomes (conditional on observable characteristics X) would have been similar for the treatment and comparison groups. This assumption would be violated if there were unobservable shocks that affected the trends in the outcomes of interest for both groups differently – shocks correlated with the policy change and with the group-specific trends in outcomes. Section 5 discusses the results from an extended version of the model of Equation 1 which includes controls for time trends interacted with the treatment group indicator. Section 6 also presents some additional specification checks to explore this possibility. An additional concern would arise if, besides the health insurance benefit extension, some other contemporaneous changes affected individuals with and without children. The introduction of HIBEX was accompanied and preceded by the implementation of a series of new programs and reforms in other existing schemes as part of an active government policy to expand the safety net in Uruguay,¹⁹ but the main results of this paper are not

¹⁸The main results are qualitatively similar when observations for 2008 are included (Table 3).

¹⁹For a detailed description of those programs and reforms, see Cruces and Bergolo (2013), Amarante et al. (2011) and Manacorda et al. (2011).

affected by the introduction of controls for participation in other programs.²⁰

The second identification assumption is that the composition of each group remained constant over the period under study.²¹ This assumption may be incorrect, for instance, if the treatment group expands over time and incorporates individuals with different characteristics. Although Equation 1 includes controls for a broad set of individual characteristics, this may not be enough to control for potential differences in group-specific compositional changes over time. Section 6 discusses in more detail these concerns and presents some sensitivity checks to test the validity of the identification strategy.

5 Empirical results

5.1 Descriptive analysis

Table 1 presents summary statistics by treatment status before and after the policy change (from the second semester of 2004 to the second semester of 2007 and from the first semester of 2009 to the second semester of 2010, respectively) for individuals in the sample with and without eligible children. The statistics in Panel A indicate that individuals in the two groups were reasonably similar in terms of age, education and other main socioeconomic characteristics. Most importantly, the last column in the table presents the difference between the pre- and post-reform changes for the two groups (effectively, an unconditional difference-in-differences analysis), which indicates only a few statistically significant changes in the average characteristics of the two groups. This evidence suggests that, at least in terms of observed characteristics, the main results discussed below are not driven by compositional effects.

The bottom rows of Panel A in Table 1 present similar descriptive statistics for the main labor market outcomes defined in Section 4.2. The average level of benefit-eligible employment before and after the reform was about 58 and 67 percent respectively for the comparison group and about 46 and 57 percent respectively for the treatment group. These differences in levels between the two groups are also reflected by the level of the not-working indicator, ranging from 21 to 16 percent for the comparison group and from 33 to 25 percent for the treatment group. On the contrary, the levels of benefit-eligible employment and of underreporting of earnings (both conditional on registered employment) were similar between the two groups for the two periods: about 91-92 percent of registered workers were eligible for the HIBEX and about 10 percent stated that their earnings were being underreported.

Figure 2, in turn, illustrates the essnee of the basic results for the outcomes of interest presented in Table 1. The Figure plots the conditional difference (with basic demographic controls) in the annual predicted benefit-eligible employment rates between 2004 and 2010

²⁰A subsample of those in the treatment group (low-income individuals with children) participated in a means tested transfer program for families with at least one dependent child ("asignaciones familiares"). As a disincentive to work, this would result in an attenuation bias for the estimates of the effects of the HIBEX. Specification checks (available upon request) indicate that this benefit did not affect the main HIBEX estimates.

²¹Abadie (2005) and Athey and Imbens (2006) discuss the implications of compositional changes in the difference-in-differences framework.

for the comparison group (childless adults) and for a subset of the treatment group, single parents with children under 18 years.²² Following the severe economic crisis that affected Uruguay in 2003-2004, the level of benefit-eligible employment increased substantially over the whole sample period for all groups. The difference in this rate between childless adults and single parents was relatively small (about 1.4 percentage points on average) and stable for the pre-reform period 2004-2007. After the reform, benefit-eligible employment grew markedly faster for single parents, which is reflected in the upward trend in the series and in the substantially larger average difference of 7.9 percentage point for 2008-2010. This simple evidence is further reinforced by the unconditional difference-in-differences estimates of the reform's effect presented in Panel A of Table 1 for the overall treatment group, which indicate an increase in benefit-eligible registered employment of about 1.6 percentage points, as well as a fall in the proportion of those not working of about 2 percentage points, with no significant differences for underreporting of earnings or for the conditional benefit-eligible variable. This preliminary evidence on the impact of the HIBEX is further developed below in terms of the full conditional difference-in-differences framework given by Equation 1.

5.2 Baseline results: Labor supply and registered employment

Table 2 presents the baseline estimates of the effects of the 2008 health insurance benefit extension on the labor market outcomes of interest discussed in Section 3. Each column reports the OLS estimates of Equation 1 for the main outcomes: benefit-eligible employment, not working, registered and unregistered employment, and benefit-employment conditional on registered work. The first row in Table 2 presents the estimates and standard errors of the interaction coefficient β in Equation 1, which captures the impact of the HIBEX. The second row displays the estimates of the coefficient of the treatment variable (*AnyChildren* in Equation 1). The last row in Table 2 reports the average of each column's dependent variable for the period before the implementation of the HIBEX (2004-2007). The discussion focuses on the specifications with the full set of controls in Equation 1; the estimates with no controls are similar in size but less precisely estimated (see Table 1, Panel A).

The estimates in column 1 of Table 2 correspond to the main outcome of interest, that is, benefit-eligible employment. The result indicates a statically significant increase in registered employment above the 25 weekly-hour cut-off for parents of young children relative to childless individuals of 1.62 percentage points. In terms of the pre-intervention average, this effect represents an increase of 3.5 percent. While small, this effect is still economically significant when compared to the trend in this variable over the period under study.

Columns 2 to 5 present the results from the additional predictions corresponding to the alternative channels of response outlined in Section 3. The negative and significant

²²The underlying specification for Figure 2 is analogous to that of Table 5, with a limited set of demographic controls and yearly treatment dummies. The sample is the same as in Table 5, panel A. This subgroup is better suited for a visual presentation of the main results because, as described below, the effects of the policy for single parents are substantially larger than for the overall treatment group.

coefficient of -1.62 for the *not working* indicator in column 2 indicates that the HIBEX induced unemployed individuals and/or those previously out of the labor force to join the labor market (a 4.9 percent increase relative to the pre-reform average). The reform could also have increased benefit-eligible employment through the formalization of unregistered employees. The positive and significant coefficient of 1.52 for registered employment in column 3 (very close to those in columns 1 and 2) and the virtually null and insignificant coefficient for the effect of the HIBEX on unregistered employment (column 4) indicate that the increase in benefit-eligible employment was mainly channeled through an increase in registered employment for those not previously working, rather than an increase in the registration of those previously in unregistered employment. The results indicate that the HIBEX did not produce, on average, a decline in unregistered employment. The distinction between single and married individuals below, however, indicates some additional effects for those facing greater changes in incentives after the reform.

Finally, the results in column 5 indicate that part of the overall effect on benefit-eligible employment was due to an increase in the hours of work above the 25 weekly-hour cut-off point for those who were registered employees.²³ This is consistent with the discussion in Section 3, since working right below the eligibility threshold becomes even less beneficial. While statistically significant at the 5 percent level, the 1.1 percentage points effect of the HIBEX for this variable is relatively small: it only implies a 1.2 percent increase relative to the pre-reform average, since about 90 percent of registered workers were above the cut-off before the policy change.

The discussion in Section 3 also implies a series of predictions about empirical results that would be incompatible with the postulated incentive effects of the HIBEX. For instance, there should be no positive effect of the reform on registered work below the hour-eligibility threshold. This placebo test can be implemented by defining the outcome as an indicator equal to one for those registered employees working less than 25 hours per week and zero for all other individuals. A regression similar to those in Table 2 yields a coefficient for the effect of the reform on this outcome of -0.001, with a standard error of 0.0037. Moreover, a further placebo test implies that the HIBEX should not have an impact on unregistered hours of work above or below 25 weekly hours - a significant difference-in-differences in this variable between the comparison and treatment groups would indicate some underlying group-specific trend and would cast doubt on the identification strategy. Conditioning the previous estimate on unregistered workers, the coefficient of the reform variable is -0.0097 and statistically insignificant at standard levels (standard error of 0.0204).

The estimates presented in Table 2 would be biased if the treatment and comparison groups had different underlying trends in labor market outcomes. Inspection of Figure 2 did not reveal major differences in the pre-reform trends for the benefit-eligible employ-

 $^{^{23}}$ A detailed analysis of hours of work for different groups indicates a substantial shift among working parents from 24 weekly hours or less to more than 25 weekly hours. Since only about 10 percent of registered workers with eligible children worked less than 25 hours and about 7 percent worked 25 to 34 hours during the pre-reform period, the sample size is too small to conduct an analysis of bunching around this threshold.

ment variable, but this can be tested formally by means of a flexible version of the basic difference-in-differences model. Table 3 presents the results from estimating Equation 1 for the full 2004-2010 period. The Table reports the coefficients of the year dummies interacted with the eligible children indicators.²⁴ The omitted period is 2008 (the year of the policy implementation), so all coefficients are measured relative to that year. Each row of the table reports the coefficients for the year-eligibility interaction for each of the dependent variables, which include the five outcomes in Table 2.

The results in Table 3 confirm the robustness of the results in Table 2 and reinforce the soundness of the identification strategy. First, only one of the 20 pre-reform year coefficients for the 5 regressions is statistically significant at the standard levels (unregistered employment in 2004), which confirms the lack of differential trends between the treatment and comparison groups in the pre-reform period. Second, the coefficient for benefit-eligible employment is large (2.49-2.6 percentage points) and statistically significant for the two years after the implementation of the policy, while that of registered employment is also relatively large (1.7-1.99 percentage points), but only significant at standard levels for 2009 (although the size of the coefficient is similar for 2010 with a p-value of about 15 percent, which suggests lack of statistical power rather than lack of effect). Third, the effect on the not-working indicator and the benefit-eligible employment conditional on registered work are only statistically significant for 2010. The lack of evidence of pre-existing differential trends and the lack of effects on the placebo outcomes support the causal interpretation of the main results in Table 2. Section 6 presents additional evidence on their robustness.

Taken together, the pattern of results in Table 2 indicates that eligible individuals responded to the health insurance benefit extension as predicted in Section 3. The bulk of the response to the additional benefit appears to have occurred due to an increase in participation rather than due to changes in hours of registered workers or shifts from off-the-book to on-the-book employment. The following section breaks the results down by subgroups facing differential incentives.

5.3 Differential incentives: Estimates by age and number of children

As discussed in Sections 2.3 and 3, the reform's financial incentives varied as a function of the age and number of an individual's dependent children. The higher cost associated with the extended benefit represented a fixed percentage of taxable earnings irrespective of the number of children who gained access to health insurance (the SIA covered the premiums of all dependent children), and thus implied a larger financial incentive for employees with more children. Moreover, younger children are associated with higher insurance and outof-pocket health expenditures and with a longer duration of the benefit. Younger children should increase the implicit value of the HIBEX for their parents.

To explore the potential heterogeneity introduced by these varying financial incentives, the basic model in Equation 1 is extended to allow for different treatment effects according

 $^{^{24}}$ The year dummies are also included independently, as described in Equation 1. The estimates in Table 3, and other tables, are based on annual rather than semester dummies because the latter are too imprecisely estimated.

to the number of dependent children and the age range of those children. The resulting specification is:

$$Y_{ijt} = \alpha + \sum_{k} \delta^{(k)} AnyChildren^{(k)}_{ijt} + \sum_{k} \beta^{(k)} AnyChildren^{(k)}_{ijt} * PostReform_t + X'_{ijt}\gamma + \theta_t + \varphi_j + \tau_{jt} + \epsilon_{ijt}$$

$$(2)$$

where $AnyChildren_{ijt}^{(k)}$ is an indicator equal to one if individual *i* in state *j* at time *t* has children in group *k*, where *k* represents one of the following four mutually exclusive groups: one child aged 0-10; one child aged 11-17; two or more children where the youngest is aged 0-10; two or more children where the youngest is aged 11-18. The matrix X_{ijt} includes the same covariates as Equation 1, and the additional terms represent the same set of state and time controls. The results from regressions based on Equation 2 are presented in Table 4. The rows in this table report the estimates of the HIBEX's impact for each number of children/age range group, $\beta^{(k)}$.

Column 1 in Table 4 indicates an increase of about 2.5 percentage points in benefiteligible employment for parents with one child aged 0-10, and of 1.8 percentage points for parents with two or more children where the youngest is aged 0-10. The coefficients for the indicators of one child and two children aged 11-17, however, are not significantly different from zero. In keeping with the literature, the greater effects in the case of parents of young children are in line with the evidence on larger employment responses to financial incentives for this group, particularly among single women (see Eissa and Liebman, 1996, and Meyer and Rosenbaum, 2001, for the EITC, and Francesconi and van der Klaauw, 2007, for the WTFC).

The pattern of results for the other labor market outcomes in Table 4 is similar: the effects are mainly driven by the indicators for having at least one child in the 0-10 age group (for registered employment, for instance), while the negative effect on the not-working indicator is mainly driven by the dummies corresponding to those with more than one child (irrespective of the age of the youngest child). Together, these results confirm the prediction of a larger reaction to the HIBEX among those with more children and among those with a longer horizon of benefits for their children, as suggested in the discussion in Section 3.

5.4 Differential incentives: Estimates by marital status

The HIBEX also introduced differential incentives for different household configurations. The incentives were stronger for single parents since, by definition, the coverage of their dependent children did not depend on the employment status of a partner and since they did not face any potential double taxation. The incentives faced by married partners, in turn, varied according to their partner's employment and benefit-eligibility status.

The results in Table 5 broadly confirm these expected differential reactions to the reform. Panel A presents the results for the same five labor market outcomes in previous tables conditioning the sample on single parents (about 17,000 individuals from a total of

more than 97,000 in the whole sample). The pattern of results is similar to the baseline estimates of Table 2, but with much larger and more significant effects, even with a substantially smaller number of observations. With respect to single childless individuals, single parents increased their level of benefit-eligible employment by 5.3 percentage points (significant at the one percent level), which represents a 14 percent increase with respect to the pre-reform average. This effect is about four times greater than the effect reported for the entire sample as shown in Table 2. The impact of the reform reduced the proportion of the not-working indicator by about 2.1 percentage points (column 2, significant at the 10 percent level) and, as such, it was similar to the effect for the entire sample. However, the estimated impact on unregistered employment (column 3) for single parents is negative and highly significant statistically, indicating a decline of about 4 percentage points in this labor market outcome indicator with respect to the comparison group. The impact of the reform on this indicator for the entire sample was virtually zero. The reductions in the number of single parents not working and of those working in unregistered employment are reflected in the sizeable (6.2 percent) increase in the number of those in registered employment (column 4, significant at the one percent level). The small and not satistically significant coefficient in column 5 (benefit-eligible employment conditional on being a registered employee) indicates that the increase in the number of single parents in benefit-eligible employment (column 1) was mainly attributable to those switching from unregistered to registered employment, with a smaller contribution from those entering the labor force and no discernible impact from registered employees increasing their hours to meet the benefit cut-off. Together, the results suggest that the positive response in the level of benefit-eligible employment for single parents after the policy change was driven by transitions from unregistered to on-the-book employment for those who were working with a smaller contribution from those entering the labor force. The smaller baseline effects for the entire sample, on the other hand, corresponded mostly to the latter factor.

Panel B in Table 5 presents estimates of the same models for the sample of married and cohabiting individuals. The estimates are substantially smaller than those for singles and in line, albeit less signifcant, than those for the entire sample (Table 2). Only the coefficient of the variable in column 5 (benefit-eligible employment conditional on being a registered employee) is statistically different from zero (and only at the 10 percent level). However, further testing indicates that this effect cannot be distinguished from the effect of the reform on the same variable for singles (p-value of 0.49).

The fact that the HIBEX had no impact on the sample of married parents, however, does not mean that there were no responses at the household level. Since the extended health coverage of a couple's children only required one of the parents to hold benefit-eligible employment, an individual's response should depend on the employment status of her partner. These household-level responses can be tested by means of an extended version of the model given by Equation 1 which includes additional indicators on whether or not the fact that an individual's partner holds a benefit-eligible job interacted with the variable capturing the HIBEX's effect (AnyChildren*PostReform). It should be stressed that the identification of these effects arises from the assumption that the eligibility status

of the spouse is exogenously given.

The estimates of these coefficients presented in Panel C in Table 5 distinguish between the two cases, labeled "Eligible partners" and "Ineligible partners." Two sets of results emerge from the estimates presented in this panel. For individuals whose partners were not in benefit-eligible employment, the HIBEX induces a statistically significant increase in this type of employment of about 1.7 percentage points (significant at the 10 percent level). This change seems to be driven mostly by an increase in registered employment, although this coefficient is not significant (column 4, coefficient 1.2). To a lesser extent, the overall change is also due to a small increase (with respect to the 0.92 pre-reform level) in benefiteligible employment for those in registered employment (column 5). On the other hand, and as expected, the HIBEX had virtually zero effect on the benefit-eligible employment (column 1) in the case of individuals whose partners received access to the extended benefit through employment. These individuals did not have the additional incentive of gaining access to the extended benefit. In fact, the reform seems to have induced a statistically significant increase in unregistered employment for this group of about 1.8 percentage points (column 4, significant at the five percent level). However, the source of this effect seems to have been a reduction in the number of those not working (column 2, coefficient of -1.5 percentage points, significant at the 10 percent level) rather than through a transition from registered to unregistered employment (column 4, coefficient of 0.27percentage points, not statistically significant), which would have been more consistent with the mechanisms highlighted in Section 3.

To sum up, the response patterns for married and cohabiting individuals are in line with the predicted effects of the reform. They are also consistent with the existing evidence on the labor supply responses to financial incentives for households (Eissa and Hoynes, 2004), and they extend this evidence to an additional margin of choice between registered and unregistered employment. These results also confirm the relevance of social insurance benefits to the observed patterns of informal employment for heads of households and their partners in Latin American countries (Galiani and Weinschelbaum, 2012).

5.5 Underreporting of salaried earnings

The results indicate that workers reacted to the reform along the registered-unregistered employment margin, with more individuals working at jobs complying with payroll taxes and contributions after the reform. This can be interpreted as an extensive margin of tax evasion - i.e., a decision of the employer and/or the employee about whether or not to evade the full amount of employment-related taxes. As the discussion in Section 3 highlights, however, changes in taxes and benefits could also have an impact on the intensive margin of evasion of employment-related taxes. Employers and/or employees might decide to pay taxes for less than the full amount due, and this type of evasion would take the form of underreporting of earnings.

The HIBEX could affect this form of non-compliance with the tax authorities and the social insurance administration in several ways. The previous results indicate that a significant number of individuals moved to benefit-eligible registered employment following the reform. To access the extended health insurance coverage for their children, these workers and their employers had to incur the full cost of payroll taxes and contributions, which amounted to more than 30 percent of taxable earnings, since the additional benefit could not be granted independently of other social insurance benefits. Moreover, for those who already held benefit-eligible registered employment, the higher contribution rate implied an incentive to increase underreporting to avoid a reduction in net earnings for workers and to avoid higher total payroll tax payments for employers.

Either as the result of collusion between employees and employees or pursuant to employers' unilateral decisions to evade taxes, the reform may have increased underreporting of salaried earnings for registered employment. As discussed in Section 4.1, the analysis of this margin of adjustment to the reform relies on workers' responses to a specific question in the survey, one that enquires about whether the contributions to the SIA for earnings from a registered job were based on the full amount of taxable earnings. Table 6, column 1, presents the difference-in-differences estimates of the HIBEX's impact on underreporting of salaried earnings to the SIA for the subsample of registered employees. As expected, the sign of the coefficient is positive, but it is not significantly different from zero. However, as discussed in Section 3, there may be heterogeneous effects on underreporting as a function of firm size. As highlighted in recent studies, tax evasion by employers and employees is more difficult for large firms to sustain, whereas it is easier for small firms to misrepresent salaried earnings (Kleven, Kreiner and Saez, 2009). The average level of underreporting of 10 percent in Table 1 varies significantly in the sample under study: in 2007, 18.5 percent of workers at firms with less than 10 employees declared that their earnings were underreported, but this figure is only 11.2 percent for those employed at firms with 10 to 49 workers and only 4.1 percent at firms with more than 50 employees.²⁵ These potential heterogeneous effects are addressed by re-estimating the same regression, but conditioning it on firm size. These results are presented in columns 2 to 4 in Table 6. Consistent with the previous discussion, the HIBEX has a positive and significant effect on underreporting only for employees at firms with less than 10 workers. With respect to those in the comparison group, the reform implied a 3.4 percentage point increase in underreporting of salaried earnings after the reform, which represents a sizeable increase (about 19 percent) in the rate of underreporting. On the other hand, the estimated effects of the reform on workers at firms with 10 to 49 workers (column 3) and on workers at firms with over 50 employees (column 4) are substantially smaller in absolute terms (-0.52)and -0.31 percentage points, respectively); in neither case are they significantly different from zero.

Finally, the reform should only increase underreporting in benefit-eligible registered employment and not in ineligible-registered employment (for those working less than the 25 weekly-hour cut-off). As an additional robustness test of these results, and to rule out the impact of other contemporaneous changes, Panels B and C in Table 6 re-estimate the same

 $^{^{25}}$ Evasion by means of unregistered employment follows a similar pattern. In 2007, the pre-reform year, 41.1 percent of employees at firms with less than 10 workers were not registered with the SIA, whereas only 9.5 percent of employees at medium-sized firms (10 to 49 employees) were not registered, and only 3.3 percent at firms with more than 50 workers.

models for all workers and by firm size, but restrict the sample to those in benefit-eligible registered employment (Panel B) and to those in benefit-ineligible registered employment (Panel C) - i.e., distinguishing between those working more and less than 25 hours per week. The only significant coefficient in the two panels is that of the effect of the reform on workers in benefit-eligible employment at small firms. In that case, the coefficient (4.1 percentage points, about 25 percent more than the pre-reform level) is similar in size to the one reported in Panel A, whereas the additional placebo test in Panel C indicates that the reform had no discernible effect (coefficient of 0.5 percentage points) on benefit-ineligible workers at small firms. The patterns of significance of the coefficients reported in these two panels confirm the previous findings: the HIBEX's impact on underreporting was confined to workers at small firms.

In sum, these results are consistent with recent empirical literature that finds evidence that changes in tax and benefit systems affect the (unilateral or collusive) decisions of firms and employees to misreport social insurance contributions and payroll taxes (Tonin, 2011; Kumler, Verhoogen and Frias, 2012). They are also in line with the prediction that this behavior is feasible at small firms but less likely on larger scales (Chetty, Friedman and Saez, 2013; Kumler, Verhoogen and Frias, 2012).

6 Robustness and specification tests

The following pages present a series of robustness and specification tests on the differencein-differences estimates presented in the previous section. These exercises are based on variations in the model of Equation 1, with full controls for individual characteristics, semester and state-fixed effects, and state-by-semester interactions effects, as in the previous analysis.

The interpretation of the difference-in-differences estimates of the HIBEX's impact as a causal effect relies on a series of assumptions. The main identifying assumption is that the changes in outcomes would have been similar for the treatment and comparison group in the absence of the reform. While these counterfactuals are inherently non-testable, it is possible to evaluate auxiliary hypotheses consistent with the identifying assumption. One option is to consider the pre-reform period and to establish whether there were common trends in the outcomes of interest, such that the reform represented a departure from the previous parallel changes.²⁶ The estimates on the year-by-year and treatment interaction terms presented in Table 3 provide a first test of the empirical strategy and establish that the difference-in-differences estimates were not simply capturing long-run differential trends between individuals with and without children.

A further concern for the identification strategy is that the treatment and comparison groups may have changed over the period under study, confounding treatment with composition effects. The summary statistics and the unconditional difference-in-differences estimates in Table 1 indicate that the main demographic characteristics of both groups

²⁶For instance, Uruguay's economy grew steadily during the period under study. Individuals with and without children might respond differently (even conditional on their observable characteristics) to episodes of growth.

did not change substantially before and after the policy change, but these cross-sectional differences in means might still conceal distributional variation across groups. For instance, since the age distribution of individuals with and without eligible children differs, the trends in their labor supply may have differed by cohort. Panel A in Table 7 presents the results of a specification that adds a full set of birth-cohort/semester interactions based on 8-year cohorts. The estimated coefficients for the main outcomes are somewhat smaller than the baseline results in Table 2, but they remain significant at the usual confidence levels, with the exception of the benefit-eligible employment conditional on registered employment outcome in column 5. The regressions were also estimated including interaction terms between the AnyChildren indicator and the full set of demographic covariates in order to allow control variables to enter separately for the treatment and the comparison groups. Results in Panel B indicate that the main estimates are also robust to this alternative. Another specification tests includes interactions of all demographic controls with the post-reform indicator to explore if other covariates could explain the responses in labor market outcomes for individuals with children relative to those without children after the HIBEX. The results in Panel C indicate that the main results are robust to the inclusion of these additional controls, with the exception of the conditional benefit-eligible employment outcome in column 5.

Finally, Panel D in Table 7 presents the estimates of the main outcomes of interest for the subsample of individuals for which there was no change in the payroll tax (those with earnings below the 2.5 BPC eligibility cut-off - see Section 2.3). As discussed above, the comparison between eligible and ineligible individuals in this group has the advantage that, while eligible employees were potential beneficiaries of the benefit extension, neither group faced an increase in payroll tax contributions. The estimated coefficients of interest are similar in magnitude to those from the baseline estimates in Table 2, although only the coefficient of the reform on the benefit-elegible employment outcome is statistically significant at the 10 percent level; the outcome on underreporting is significant at the 5 percent level. Since this group did not see an increase in payroll tax, we should expect a larger effect, and this is in fact the case: relative to the pre-treatment dependent variable mean of 0.46, the effect on eligible employment for the whole sample of 0.0162 (Table 2, column 1) represents an increase of about 3.5 percent. The corresponding pre-treatment mean is 0.11, so the 0.0168 increase (Table 7, panel D, column 1) implies a much greater effect in relative terms (about 15.3 percent greater). The fact that the estimates are still robust for this subgroup reinforces the identification strategy, indicating that the increase in payroll tax for the control group is not a concern. The analysis for this subgroup, however, lacks statistical power: the samples for these alternative comparison and treatment groups are substantially smaller and, unlike the analysis for the full sample in Tables 4 and 5, this one does not allow for a meaningful analysis by subgroup.

In general, these robustness and specification tests suggest that changes in the composition of the treatment and comparison groups did not introduce a spurious correlation between changes in the outcomes before and after the reform.

7 Financial implications and discussion

The HIBEX was designed to increase the scope of health insurance coverage for the dependent children of registered employees. As depicted in Figure 1, take-up of the benefit was very high, triggering a large increase in government expenditure in the form of private health insurance premiums for newly eligible children. Part of this expense was offset by a reduction in the costs of the public health system. In terms of revenues, the reform had several implications. On the one hand, tax collection increased. An increase in contributions for most registered workers generated higher payroll tax receipts. On the other hand, in addition to this direct or mechanical effect, the reform also resulted in an increase in registered employment. This behavioral effect generated both higher tax collection and higher expenditures. Finally, the results discussed above also indicate that the reform had a dual effect on tax evasion in the labor market: registered employment increased, but underreporting of salaried earnings increased as well. The following pages discuss the overall impact of the reform on government finance due to each of these effects, which are summarised in Table 8. This estimate of the fiscal incidence of the reform is restricted to individuals in the survey sample, and it relies on a series of assumptions.²⁷

The first step for computing the fiscal incidence of the reform is to estimate the mechanical changes in tax revenue and expenditure. Childless registered workers did not imply any additional expenditure after the reform, but they represented higher revenues. The fraction of the payroll tax allocated to health insurance for these workers increased by 1.5 percentage points. Based on an average monthly salary of UYU 12,889.2 for the 67,122 childless registered employees working 25 hours or more per week in 2007, annual government revenue increased by UYU 155.7 million from this source (row A.1, column 6 - all values in monthly UYU are multiplied by 12 to obtain the results in annual terms).²⁸

The second mechanical effect follows the change in payroll taxes and healthcare insurance benefits for the 159,250 individuals with children in benefit-eligible employment in the pre-reform year, 2007. In terms of revenue, based on the average taxable earning of UYU 12,689.2 per month in 2007 for this group, the 3 percentage point increase in health insurance contributions resulted in an increase in revenue of UYU 727.5 million (row A.2, column 6). On the expenditure side, the reform generated an additional expenditure in the form of health insurance premiums paid by the SIA for the children of these workers in 2008 at UYU 603 per child per month (in 2007 prices). The total number of children covered by the reform was 242,072, which resulted in an additional expenditure of UYU 1,750,7 million (row D.1, column 6). However, a substantial fraction of the

²⁷The values of all parameters correspond to the year 2007, except when indicated otherwise (for instance, the parameters that changed with the reform (contributions) or new parameters, such as the insurance premiums paid by the SIA, which are based on figures from 2008). Information on expenditure per individual, such as health insurance premiums, are taken from administrative sources (Banco de Previsión Social, 2008; 2009; Ministerio de Salud Pública, 2010a). All monetary values in UYU are expressed in 2007 prices.

²⁸As a further simplifying assumption, the analysis does not take into account the level of underreporting for those already in registered employment before the reform (rows A.1 and A.2), which stayed roughly constant. This would complicate the analysis and require further assumptions, which would, in turn, result in somewhat lower figures for tax revenue in column 6 of rows A.1 and A.2. The analysis highlights instead the reform-induced behavioral changes in underreporting of earnings for workers at small firms (row B.2).

children were covered by the public health sector in 2007 at an average cost of UYU 350 per child per month, so this shift in their health insurance arrangements from public to private amounted to savings of UYU 683.1 million (row D.2, column 6) and resulted in a substantially lower net increase in expenditures.

The changes discussed thus far are limited to mechanical effects. The evidence in this paper, however, indicates the presence of behavioral effects due to the HIBEX, which has further consequences for government expenditure and tax revenues. The estimates from Table 2 show that, on average, the HIBEX increased benefit-eligible employment by 1.62 percentage points, a 3.5 percent increase with respect to the pre-reform levels of 2007. This represented an increase of 5.574 (0.035×159.250) in the number of workers in benefit-eligible employment and 8,473 newly eligible children (average of 1.52 children per registered employee in 2007). Computing the annual premiums for adults (UYU 809) and children (UYU 604) paid by the SIA in 2008, the additional annual expenditure in health insurance attributable to this behavioral effect is UYU 115.4 million (rows E.1 and E.2, column 6). However, 4,450 of the newly covered adults and 5,692 of their children were covered by the public sector before the reform (based on the 2007 coverage of the public healthcare system, 67 percent for parents and 81.5 percent for children), so this shift in healthcare insurance arrangements from public to private reduced public-sector expenditure by UYU 350 per individual per month for a total of UYU 43 million (row E.3, column 6).

In terms of revenue, the increase of 5,574 in benefit-eligible registered workers implied higher contributions to the health insurance fund for an annual total of UYU 93.4 (row B.1), yielded by an average monthly earning per worker of UYU 12,689.2, the additional pre-reform employer contributions (5 percent of earnings), the additional pre-reform employee contribution (3 percent), and the increase in the payroll tax rate due to the reform for this group (an additional 3 percent of earnings).

Finally, also in terms of revenue, an additional factor is the increase in underreporting of salaried earnings for registered employees at small firms. It can be assumed that earnings were underreported by 25 percent, which represents the middle point between average earnings for registered employees at small firms and the minimum wage. Table 6, Panel B, indicates that the reform increased underreporting by 4.09 percentage points for benefit-eligible employees at small firms, which represented an increase of 24 percent in the incidence of underreporting with respect to the pre-reform period. Of a total of 45,042 registered employees with children working at small firms in 2007, 7,657 (17 percent) declared underreported earnings, and the average monthly earning for this group in 2007 was UYU 8,231. Taking into account the 3 percent increase in contributions, tax revenue from this effect fell by UYU 1.4 million. A similar analysis applies to the subset of parents who entered benefit-eligible employment in small firms after the reform (about 28 percent of the original 5,574 workers). A total of 332 of these 1,576 new workers underreported earnings. Including the original health insurance contribution of 8 percent and the additional 3 percent due to the reform, this implied an additional fall in (potential) tax collection of UYU 0.9 million. Overall, tax collection fell by UYU 2.3 million due to the behavioral response of underreporting salaried earnings (row B.2, column 6).²⁹

To sum up, on the basis of these assumptions, the direct or mechanical effect of the reform implies an increase in the budget deficit of ([1,750.7-683.1] - [155.7+727.5]) = UYU 184.5 million. Furthermore, the behavioral impact of the reform actually implied an additional total effect of ([54.1+61.3-43.0] - [93.4-2.3]) = UYU -18.7 million, i.e., a reduction in the budget deficit. This implies that the increase in tax collection through the increase in the number of registered workers more than compensated for the additional public expenditure incurred by covering the health insurance premiums of those individuals and their families. Another relevant conclusion is that the additional revenue created by the increase in registered employment following the reform, UYU 93.4 million, exceeded by a factor of about 40 the loss of revenue of UYU 2.3 million due to the higher levels of underreporting induced by the HIBEX; the increase in the intensive margin of payroll tax evasion was of a second order when compared to the gains from the reduction in its extensive margin.

Taken together, the results indicate that the HIBEX increased the budget deficit by (1,140.1-974) = UYU 165.8 million or USD 10.5 million at the 2010 PPP adjusted exchange rate, which represents 0.030 percent of Uruguay's GDP in 2007 and 0.28 percent of total government expenditure on health care for that year. This additional 0.030 percent of GDP more than tripled the coverage of explicit health insurance for children.³⁰ While the specific figures depend on a series of assumptions, they provide suggestive evidence on the relative magnitude of the different effects at play. There are probably other important positive effects in terms of children's health and other spillovers from registered employment (access to pensions and other benefits), which should be included in a full cost-benefit analysis which is beyond the scope of this paper.

8 Conclusions

This paper studied the impact of a reform in Uruguay's social insurance system that extended the coverage of a health insurance benefit from registered employees to their dependent children. This extended benefit introduced a new incentive for parents to take up registered employment. The evidence confirms that individuals reacted as predicted, with a substantial increase in benefit-eligible employment for parents as compared to childless individuals. Also as predicted, single parents, parents with several children, and parents with younger children entered registered employment at a higher rate. For individuals

²⁹This total is yielded by the UYU 1.4 million $(0.24 \times 7,657 \times 0.25 \times 8,231 \times 0.03)$ and the 332 $(1,572 \times 0.17 \times [1+0.24])$ new underreporters for UYU 0.9 million $(332 \times 0.25 \times 8,231 \times 0.11)$.

 $^{^{30}}$ The total number of children enrolled in the program after the reform in 2008 was 250,545 (Table 8, D.1+E.2). A first part of this total corresponds to the 79,440 children of registered employees who were already enrolled in private health insurance because their parents paid for it before the reform. The second component corresponds to the mechanical effect for children of registered employees. Before the reform, there were 162,632 children of registered employees who used the public health sector, and who then enrolled in private health insurance because of the reform (row D.2). The third component is the behavioral response. Parents who entered benefit eligible employment and enrolled their eligible children implied 8,473 newly covered children (row E.1). Compared to the baseline coverage of 79,440 children, the behavioral response represented an increase of about 10.7% in the coverage of explicit health insurance.

in couples, responses to the reform depended on their partner's employment's benefiteligibility. The increase in benefit-eligible employment was driven by both reductions in the number of those not working and (for single parents) by shifts from unregistered to registered employment. The results indicate a complex pattern of tax-evasion responses in the labor market following the policy change. The reform also induced reactions along a further margin of adjustment in tax compliance, with higher levels of underreporting of taxable earnings for benefit-eligible registered employees working at small firms. A fiscal incidence analysis of the reform indicates that the payroll tax increases associated with the benefit extension managed to cover most but not all of the increase in expenditure. Moreover, the reform's behavioral effect amounted to a small, but relevant, fraction of the total change in the budget deficit. Finally, the increase in revenue from the reform's positive effects on registered employment far outweighed the loss in revenue from increased underreporting of salaried earnings.

The analysis indicates that additional tax evasion margins of response mean complex trade-offs for workers and governments in the context of tax and benefit systems. In terms of tax evasion, the decision is not simply between off-the-book versus registered employment, but between full non-compliance, full compliance, and on-the-book salaried employment with under-the-table payments. These results confirm the insight that work incentives do not necessarily operate within the framework of the law and that the design of social insurance systems should take into account these additional dimensions.

The implications of this analysis are relevant not only for developing economies with large informal labor forces. The policy consequences are also relevant for OECD countries, most of which have experienced an increase in their shadow economies over the past two decades (Schneider and Enste, 2000; Schneider, 2005); they may well face trade-offs similar to those described for Uruguay, at least for part of their labor forces. Moreover, it is necessary to understand the channels through which the tax and benefit system yields changes in evasion behavior in the context of salaried employment. For instance, it would be important to distinguish between situations where the employer and the employee collude to evade taxes from settings where the firm engages in unilateral evasion. Thirdparty information reporting (which has been highlighted as an effective measure to induce compliance) and other specifics of tax law and implementation details may also play an important role in the reduction of unregistered employment and of the informal sector in developing countries.

References

- Abadie, A., 2005. Semiparametric difference-in-differences estimators. The Review of Economic Studies 72 (1), 1-19.
- Allingham, M. G., Sandmo, A., 1972. Income tax evasion: a theoretical analysis. Journal of Public Economics 1(3-4), 323-338.
- Amarante, V., Manacorda, M., Miguel, E., Vigorito, A., 2011. Do cash transfers improve birth outcomes? Evidence from matched vital statistics, social security and program data. NBER Working Paper 17690.
- Angrist, J. D., Krueger, A. B., 1999. Empirical strategies in labor economics. In: Ashenfelter, Orley, Card, David (Eds.), Handbook of Labor Economics, Volume 3A. Elsevier, Amsterdam, pp. 1277-1366.
- Athey, S., Imbens, G. W., 2006. Identification and inference in nonlinear difference-indifferences models. Econometrica 74(2), 431-497.
- Banco de Previsión Social, 2008. Boletín Estadístico 2008, BPS, Montevideo. Available at: http://www.bps.gub.uy/BrowserNetCM.aspx?menu=institucional&res=Institucional%2festadisticas
- Banco de Previsión Social, 2009. Boletín Estadístico 2009, BPS, Montevideo. Available at: http://www.bps.gub.uy/BrowserNetCM.aspx?menu=institucional&res=Institucional%2festadisticas
- Ben-Shalom, Y., Moffitt, R., Scholz, J. K., 2011. An Assessment of the Effectiveness of Anti-Poverty Programs in the United States. In: Jefferson, P. (Ed.), Oxford Handbook of the Economics of Poverty, Oxford University Press.
- Bergolo, M., Cruces, G., 2013. Labor Informality and the Incentive Effects of Social Protection Systems: Evidence from a Health Reform in Uruguay. Mimeo, IZA and IDB.
- Blundell, R., MaCurdy, T., 1999. Labor supply: a review of alternative approaches. In: Ashenfelter, O., Card, D. (Eds.), Handbook of Labor Economics, Volume 3A. Elsevier Science, Amsterdam, pp. 1559-1695.
- Borgia, F., 2008. Health in Uruguay: Progress and Challenges in the Right to Health Care Three Years after the First Progressive Government. Social Medicine 3(2), 110-125.
- Bound, J., Burkhauser, R. V., 1999. Economic analysis of transfer programs targeted on people with disabilities. In: O. Ashenfelter and D. Card (Eds.), Handbook of Labor Economics, Volume 3. Elsevier, Amsterdam, pp. 3417-3528.
- Bosch, M., Manacorda, M., 2012. Social Policies and Labor Market Outcomes in Latin America and the Caribbean: A Review of the Existing Evidence. CEP Occasional paper 32, LSE.
- Chetty, R., 2009. Is the Taxable Income Elasticity Sufficient to Calculate Deadweight Loss? The implications of Evasion and Avoidance. American Economic Journal: Economic Policy 1(2), 31-51.
- Chetty, R., Friedman, J. N., Saez, E., 2013. Using Differences in Knowledge Across Neighborhoods to Uncover the Impacts of the EITC on Earnings. American Economic Review, 103(7): 2683-2721..
- Cowell, F. A., 1985. Tax Evasion with Labour Income. Journal of Public Economics 26 (1), 19-34.
- Cruces, G., Galiani, S., Kidyba, S., 2010. Payroll taxes, wages and employment: Identification through policy changes. Labour Economics 17(4), 743-749.

- Cruces, G., Bergolo, M., 2013. Informality and Contributory and Non-Contributory Programmes. Recent Reforms of the Social Protection System in Uruguay. Development Policy Review 31(5).
- Eissa, N., Liebman, J. B., 1996. Labor Supply Response to the Earned Income Tax Credit. Quarterly Journal of Economics 111(2), 605-637.
- Eissa, N., Hoynes, H., 2004. Taxes and the labor market participation of married couples: the earned income tax credit. Journal of Public Economics 88(9-10), 1931-1958.
- Eissa, N., H., Hillary W., 2006. Behavioral Responses to Taxes: Lessons from the EITC and Labor Supply. In: J. M. Poterba (Ed.), Tax Policy and the Economy, Vol. 20, MIT Press, Cambridge, pp. 73-110.
- Feldman, N. E., Slemrod, J., 2007. Estimating tax noncompliance with evidence from unaudited tax returns. The Economic Journal 117(518), 327-352.
- Ferreira-Coimbra, N., Forteza, A., 2004. Protección social en Uruguay: Financiamiento, cobertura y desempeño, 1990-2002. Oficina Internacional del Trabajo, Santiago de Chile.
- Francesconi, M., van der Klaauw, W., 2007. The socioeconomic consequences of "in-work" benefit reform for British lone mothers, Journal of Human Resources 42(1), 1-31.
- Galiani, S., Weinschelbaum, F., 2012. Modelling Informality Formally: Households and Firms. Economic Inquiry 50(3), pp. 821-838.
- Gasparini, L., Tornarolli, L., 2009. Labor Informality in Latin American and the Caribbean: Patterns and Trends from Household Surveys Microdata. Desarrollo y Sociedad 63 (June), 13-80.
- Gruber, J., 1997. The Incidence of Payroll Taxation: Evidence from Chile. Journal of Labor Economics 15(S3), S72-S101.
- Kleven, H. J., Kreiner, C. T., Saez, E., 2009. Why can Modern Governments Tax so Much? An Agency Model of Firms as Fiscal Intermediaries. NBER Working Papers 15218.
- Kleven, H.. J., Knudsen, M., Kreinerm, C. T., Pedersen, S. L., Saez, E., 2011. Unwilling or Unable to Cheat? Evidence from a Tax Audit Experiment in Denmark. Econometrica 79 (3), 651-692.
- Krueger, A. B., Meyer, B. D., 2002. Labor supply effects of social insurance. In: Auerbach, Alan, Feldstein, Martin (Eds.), Handbook of Public Economics, Volume 4. North-Holland, Amsterdam, pp. 2327-2392.
- Kumler, T. J., Verhoogen, E. A., Frias, J., 2012. Enlisting Workers in Monitoring Firms: Payroll Tax Compliance in Mexico. Columbia University Department of Economics Discussion Papers 96.
- La Porta, R., Shleifer, A., 2008. The Unofficial Economy and Economic Development. Brookings Papers on Economic Activity, 2008, pp. 275-352.
- Levy, S., 2008. Good Intentions, Bad Outcomes: Social Policy, Informality and Economic Growth in Mexico. The Brookings Institute, Washington, D.C.
- Manacorda, M., Miguel, E., Vigorito, A., 2011. Government Transfers and Political Support. American Economic Journal: Applied Economics 3 (3), 1-28.
- Meyer, B. D., Rosenbaum, D. T., 2001. Welfare, the earned income tax credit, and the labor supply of single mothers. Quarterly Journal of Economics 116 (3), 1063-1114.

- Ministerio de Salud Pública, 2010a. Cuentas Nacionales en Salud 2005-2008. MSP, Montevideo. Available at http://www.msp.gub.uy/ucecsalud_5734_1.html
- Ministerio de Salud Pública, 2010b. Logros y desafíos en términos de Equidad en Salud en Uruguay, Documentos de Trabajo de Economía de la Salud No 1/10, MSP, Montevideo. Available at http://www.msp.gub.uy/ucecsalud_5680_1.html
- Moffitt, R. A., 2003. The temporary assistance to needy families program. In: Moffitt, Robert A. (Ed.), Means-Tested Transfer Programs in the United States, NBER and University of Chicago Press, Chicago, pp. 291-363.
- Potter Gunter, S., 2013. State Earned Income Tax Credits and the Participation in Regular and Informal Work. ational Tax Journal, vol. 66(1), p. 33-62.
- Saez, E., Slemrod, J., Giertz, S., 2012. The Elasticity of Taxable Income with Respect to Marginal Tax Rates: A Critical Review. Journal of Economic Literature 50(1), 3-50.
- Sandmo, A., 1981. Income Tax Evasion, Labour Supply, and the Equity-Efficiency tradeoff. Journal of Public Economics 16(3), 265-288.
- Sandmo, A., 2012. An evasive topic: theorizing about the hidden economy. International Tax Public Finance 19(1), 5-24.
- Schneider, F., 2005. Shadow economies around the world: what do we really know?. European Journal of Political Economy 21 (3), 598-642.
- Schneider, F., Enste, Dominik H., 2000. Shadow economies: size, causes, and consequences. Journal of Economic Literature 38 (1), 77-114.
- Shaw, J., Slemrod, J., Whiting, J., 2010. Administration and Compliance. In: Besley, Adam, S., Besley, T. Blundell, R., Bond, S., Chote, R., Gammie, M., Johnson, P., Myles, G., Poterba, J. (Eds.), Dimensions of Tax Design: The Mirrlees Review. Institute of Fiscal Studies and Oxford University Press, pp. 1100–1162.
- Slemrod, J., Yitzhaki, S., 2002. Tax Avoidance, Evasion and Administration. In: Auerbach, A. J., Feldstein, M. (Eds), Handbook of Public Economics vol. 3. Elsevier, Amsterdam.
- Slemrod, J., 2007. Cheating ourselves: The economics of tax evasion. Journal of Economic Perspectives, 21(1), 25-48.
- Slemrod, J., Weber, C., 2012. Evidence of the Invisible: Toward a Credibility Revolution in the Empirical Analysis of Tax Evasion and the Informal Economy. International Tax and Public Finance 19 (1), 25-53.
- Tonin, M., 2011. Minimum wage and tax evasion: Theory and evidence. Journal of Public Economics 95 (11-12), 1635-1651.
- Yitzhaki, S., 1974. A note on income tax evasion: a theoretical analysis. Journal of Public Economics 3(2), 201-202.



Figure 1: Healthcare coverage in the private market before and after the reform (2004-2010).

Note: The sample corresponds to the Encuesta Continua de Hogares from the 2nd semester of 2004 to the 2nd semester of 2010.

Figure 2: Difference in benefit-eligible employment rates (registered and working 25 hours or more per week) by year with demographic controls. Single parents with children under the age of 18 and childless adults.



Note: The sample contains individuals aged 25 to 55 and corresponds to the Encuesta Continua de Hogares (ECH) from the 2nd semester of 2004 to the 2nd semester of 2010. The set of controls includes the individual's age, gender, head of household status, education level, number of children, head of household status and marital status.

		hildless	individual	s	Pare	nts with	children <	< 18	Difference in
	Pre-re	eform	Post-r	eform	Pre-re	eform	Post-re	eform	
	Mean	SD	Mean	SD	Mean	SD	Mean	SD	differences
Panel A. Descritptive statistics									
Sample									
N (Total=97,552)	13,402		9,756		45,973		28,421		
Share	0.14		0.10		0.47		0.29		
Individual characteristics									
Age	39.66	9.93	39.62	9.89	38.49	7.62	38.56	7.58	0.117
High school or less	0.72	0.45	0.72	0.45	0.85	0.35	0.86	0.35	0.002
Some college or more	0.28	0.45	0.28	0.45	0.15	0.35	0.14	0.35	-0.002
Male	0.48	0.50	0.49	0.50	0.39	0.49	0.40	0.49	0.005
Head of household	0.63	0.48	0.65	0.48	0.50	0.50	0.53	0.50	0.012
Married	0.65	0.48	0.63	0.48	0.89	0.32	0.87	0.33	0.013*
Number of children aged 0 - 10	0.00	0.00	0.00	0.00	1.12	0.99	1.06	0.95	-0.061**
Number of children aged 11 - 17	0.00	0.00	0.00	0.00	0.80	0.83	0.81	0.85	0.013*
Number of children > 18	0.00	0.00	0.00	0.00	0.28	0.60	0.26	0.57	-0.016**
Total number of children	0.00	0.00	0.00	0.00	2.20	1.13	2.14	1.10	-0.064**
Region									
Montevideo	0.71	0.46	0.69	0.46	0.59	0.49	0.60	0.49	0.024***
North	0.06	0.24	0.06	0.24	0.10	0.30	0.09	0.29	-0.007*
Center - North	0.07	0.25	0.08	0.27	0.10	0.30	0.10	0.30	-0.007
Center - South	0.05	0.23	0.06	0.23	0.07	0.26	0.08	0.26	-0.002
South	0.11	0.32	0.11	0.32	0.13	0.34	0.13	0.33	-0.007
Firm size									
< 10 employees	0.41	0.49	0.34	0.47	0.44	0.50	0.38	0.49	0.014
10-49 employees	0.22	0.42	0.24	0.43	0.22	0.42	0.23	0.42	-0.009
> 49 employess	0.36	0.48	0.43	0.49	0.34	0.47	0.39	0.49	-0.005
Sector									
Agriculture	0.04	0.20	0.04	0.20	0.06	0.24	0.06	0.23	0.001
Industry	0.10	0.30	0.09	0.28	0.11	0.32	0.11	0.31	0.005
Manufacturing	0.07	0.26	0.08	0.28	0.08	0.26	0.07	0.26	-0.010**
Construction	0.06	0.23	0.06	0.25	0.07	0.26	0.09	0.28	0.005
Trade	0.23	0.42	0.23	0.42	0.20	0.40	0.22	0.41	0.016**
Transport/communications	0.07	0.26	0.09	0.28	0.08	0.27	0.08	0.27	-0.012**
Finance/professional	0.11	0.32	0.12	0.32	0.08	0.26	0.07	0.26	-0.009
Education/health	0.19	0.40	0.18	0.39	0.17	0.38	0.17	0.37	0.005
Personal services	0.12	0.32	0.10	0.31	0.15	0.36	0.14	0.34	-0.001
Labor market and tax evasion outo	comes								
Benefit-eligible employment	0.58	0.49	0.67	0.47	0.46	0.50	0.57	0.50	0.016*
Not working	0.21	0.41	0.16	0.36	0.33	0.47	0.25	0.43	-0.020**
Unregistered employment	0.16	0.36	0.12	0.32	0.16	0.37	0.13	0.34	0.005
Registered employment	0.63	0.48	0.72	0.45	0.51	0.50	0.62	0.49	0.014*
Benefit-eligible empl. reg. empl.	0.92	0.26	0.92	0.26	0.91	0.29	0.92	0.28	0.007
Underreporting earnings to SIA	0.10	0.30	0.09	0.28	0.10	0.30	0.10	0.30	0.01
Panel B. Payroll tax contributions to	the Social S	Security	Administ	ration					
Health insurance benefit	8	-	9.5		8		11		
Employee	3		4.5		3		6		
Employer	5		5		5		5		
Other social insurance benefits	22.75		22.75		22.75		22.75		
Employee	15.125		15.125		15.125		15.125		
Employer	7.625		7.625		7.625		7.625		
Total	30.75		32.25		30.75		33 75		

Table 1: Summary statistics (Encuesta Continua de Hogares, ECH) 2004-2010 and payroll tax contributions for social insurance benefits.

Notes: SD denotes the standard deviation. "Pre-reform" refers to the period from the 2nd semester of 2004 to the 2nd semester 2007. "Post-reform" denotes the period from the 1st semester of 2009 to the 2nd semester of 2010. "Difference in differences" refers to the impact coefficient estimated by running a regression as in Equation 1 (without controls) on each variable in Panel A with Huber-White robust standard errors. "Parents with children <18" includes all adults in the relevant age range with at least one child under the age of 18 who may or may not have older children. The sample includes individuals aged 25-55 and contemplates the variables of benefit-eligible employment, not working, unregistered and registered employment (97,552 observations); only those working in registered employment are included for benefit-eligible employment conditional on registered employees aged 25-55 from 2006 onwards (50,669 observations). Means are weighted by ECH sampling weights. The payroll tax contributions correspond to December 2007 ("Pre-reform") and 2008 ("Post-reform"). The "Health insurance benefit" includes contributions for both sickness benefit and health insurance coverage. The change in contributions - UYU 1,775 in 2008). Payroll tax contributions for registered workers with salaried earnings below 2.5 BPC remained unchanged after the reform. "Other social insurance benefits" pension contributions and contributions to the "Fondo de Reconversión Laboral" (labor reallocation fund).

* significant at 10%; ** significant at 5%; *** significant at 1%.

	(1)	(2)	(3)	(4)	(5)
	Benefit-eligible employment	Not working	Unregistered employment	Registered employment	Benefit-eligible employment reg. empl.
AnyChildren*PostReform	0.0162**	-0.0162**	0.001	0.0152**	0.0107**
	[0.0075]	[0.0063]	[0.0058]	[0.0073]	[0.0052]
AnyChildren	0.0239***	-0.0292***	-0.0052	0.0344***	-0.0132***
	[0.0066]	[0.0059]	[0.0055]	[0.0066]	[0.0048]
Observations	97,552	97,552	97,552	97,552	56,707
R^2	0.26	0.24	0.05	0.25	0.17
Dependent variable mean	0.46	0.33	0.16	0.51	0.91

Table 2: Effect of the health insurance benefit extension on labor market outcomes. Difference in differences estimates.

Notes: The sample includes individuals aged 25-55 (all individuals for columns 1-4, conditional on registered employment for column 5). The data corresponds to the Encuesta Continua de Hogares (ECH) from the 2nd semester of 2004 to the 2nd semester 2007 and from the 1st semester of 2009 to the 2nd semester of 2010. "Benefit-eligible employment" refers to registered employment with 25 hours or more per week. The "AnyChildren" variable equals 1 if the individual has at least one child under the age of 18 and 0 otherwise. The "Post-reform" variable equals 1 for the period 2009-2010 and 0 otherwise. The estimated coefficients correspond to the regression in Equation 1. The regressions are estimated as linear probability models. The controls include the number of children; the number of children aged 0-5; dummy variables for: age (6 categories), gender, head of the household status, marital status, educational level (6 categories), region of residence (19 categories), firm size (5 categories), industry (9 categories) (the final two are only included in the benefit-eligible registered employment conditional on registered employment regression), and semester (11 categories). The controls also include interactions between age and education groups and number of children, and time and region of residence. Regressions are weighted by ECH sampling weights. The reported means for the dependent variables correspond to the pre-reform period (2004-2007).

	(1)	(2)	(3)	(4)	(5)
	Benefit-eligible employment	Not working	Unregistered employment	Registered employment	Benefit-eligible employment reg. empl.
AnyChildren*Year di	ummies				
2010	0.0249**	-0.0222**	0.0053	0.017	0.0158**
	[0.0114]	[0.0095]	[0.0084]	[0.0110]	[0.0077]
2009	0.0260**	-0.0064	-0.0135	0.0199*	0.0106
	[0.0114]	[0.0094]	[0.0084]	[0.0110]	[0.0076]
2007	0.0012	0.0009	0.0024	-0.0033	0.0053
	[0.0110]	[0.0091]	[0.0083]	[0.0107]	[0.0075]
2006	0.0087	-0.0061	-0.0014	0.0075	-0.0047
	[0.0111]	[0.0094]	[0.0083]	[0.0109]	[0.0076]
2005	0.0173	0.0065	-0.0083	0.0018	0.0154
	[0.0147]	[0.0128]	[0.0122]	[0.0145]	[0.0109]
2004	0.0211	0.0222	-0.0339**	0.0118	0.0039
	[0.0194]	[0.0168]	[0.0166]	[0.0192]	[0.0139]
Observations	117,001	117,001	117,001	117,001	68,515

Table 3: Effect of the health insurance benefit extension on labor market outcomes. Dynamic specification of the difference in differences model.

Notes: The sample includes individuals aged 25-55 (all individuals for columns 1-4, conditional on registered employment for column 5). The data corresponds to the Encuesta Continua de Hogares (ECH) from the 2nd semester of 2004 to the 2nd semester of 2010. Unlike all the other tables in the paper, the estimates presented here include the two semesters of 2008, the year of the reform. The omitted year category is 2008. "Benefit-eligible employment" refers to registered employment with 25 hours or more per week. The "AnyChildren" variable equals 1 if the individual has at least one child under the age of 18 years and 0 otherwise. The "Post-reform" variable equals 1 for the period 2009-2010 and 0 otherwise. The estimated coefficients correspond to the regression in Equation 1. The regressions are estimated as linear probability models. The controls include the number of children; the number of children aged 0-5; dummy variables for: age (6 categories), gender, head of the household status, marital status, education (6 categories), region of residence (19 categories), firm size (5 categories), industry (9 categories) (the final two are only included in the benefit-eligible registered employment conditional on registered employment regression), and semester (11 categories). The controls also include interactions between age and education groups and number of children, time and region of residence, and AnyChildren and year dummies. Regressions are weighted by ECH sampling weights. The reported means for the dependent variables correspond to the pre-reform period (2004-2007).

Table 4: Effect of the health insurance benefit extension on labor market outcomes by age and number of children. Difference in differences estimates.

	(1)	(2)	(3)	(4)	(5)
	Benefit-eligible employment	Not working	Unregistered employment	Registered employment	Benefit-eligible employment reg. empl.
Child group dummies*Po	stReform				
One child only, aged 0-10	0.0246**	-0.0126	-0.0119	0.0245**	0.0066
	[0.0100]	[0.0085]	[0.0075]	[0.0098]	[0.0069]
One child only, aged 11-17	0.0084	-0.012	0.0026	0.0095	0.0113
	[0.0104]	[0.0092]	[0.0081]	[0.0103]	[0.0078]
Two children or more,	0.0183**	-0.0184**	0.0031	0.0153*	0.0116*
youngest aged 0-10	[0.0083]	[0.0073]	[0.0066]	[0.0082]	[0.0060]
Two children or more,	0.0023	-0.0212*	0.0174	0.0038	0.0123
youngest aged 11-17	[0.0133]	[0.0119]	[0.0110]	[0.0132]	[0.0103]
Observations	97,552	97,552	97,552	97,552	56,707
Dependent variable mean	0.46	0.33	0.16	0.51	0.91

Notes: The sample includes individuals aged 25-55 (all individuals for columns 1-4, conditional on registered employment for column 5). The data corresponds to the Encuesta Continua de Hogares (ECH) from the 2nd semester of 2004 to the 2nd semester 2007 and from the 1st semester of 2009 to the 2nd semester of 2010. "Benefit-eligible employment" refers to registered employment with 25 hours or more per week. The dependent variables and controls are the same as those detailed in the notes to Table 2. The regressions are estimated as linear probability models. The rows report the coefficients estimated for each number–by–age range of children group from regressions as in Equation 2. Regressions are weighted by ECH sampling weights. The reported means for the dependent variables correspond to the pre-reform period (2004-2007).

	(1)	(2)	(3)	(4)	(5)
	Benefit-eligible employment	Not working	Unregistered employment	Registered employment	Benefit-eligible employment reg. empl.
Panel A. Singles					
AnyChildren*PostReform	0.0530***	-0.0207*	-0.0417***	0.0624***	0.0017
	[0.0158]	[0.0124]	[0.0137]	[0.0153]	[0.0127]
Observations	17,340	17,340	17,340	17,340	10,387
Dependent variable mean	0.38	0.27	0.29	0.44	0.86
Panel B. Individuals with p	artners				
AnyChildren*PostReform	0.0102	-0.0112	0.0047	0.0064	0.0114*
	[0.0088]	[0.0078]	[0.0065]	[0.0086]	[0.0062]
Observations	80,212	80,212	80,212	80,212	46,320
Dependent variable mean	0.47	0.33	0.15	0.52	0.92
P-value singles = couples	0.0177**	0.5173	0.0020***	0.0014***	0.4991
Panel C. By partner's emplo	oyment status				
Reform's effect on those	0.0168*	-0.0082	-0.0035	0.0117	0.0125*
with ineligible partners	[0.0093]	[0.0082]	[0.0070]	[0.0091]	[0.0065]
Reform's effect on those	-0.0012	-0.0155*	0.0182**	-0.0027	0.0090
with eligible partners	[0.0099]	[0.009]	[0.0071]	[0.0097]	[0.0071]
Observations	80,212	80,212	80,212	80,212	46,320
Dependent variable mean	0.47	0.33	0.15	0.52	0.92

Table 5: Effect of the health insurance benefit extension on labor market outcomes of individuals by marital status and partner's employment status. Difference in differences estimates.

Notes: The sample includes individuals aged 25-55 (all individuals for columns 1-4, conditional on registered employment for column 5). The data corresponds to the Encuesta Continua de Hogares (ECH) from the 2nd semester of 2004 to the 2nd semester of 2007 and from the 1st semester of 2009 to the 2nd semester of 2010. "Benefit-eligible employment" refers to registered employment with 25 hours or more per week. The dependent variables and controls are the same as those detailed in the notes to Table 2. The regressions are estimated as linear probability models. The specification is similar to that of Equation 1 with two exceptions: it includes an indicator for the employment status of the individual's partner and its interaction in order to capture the reform's impact (AnyChildren*PostReform). Panels A and B report estimates from a regression as Equation 1 for single individuals conditioning on their partner's employment status, that is, holding a benefit-eligible registered job ("Partner eligible") or not ("Partner ineligible"). Regressions are weighted by ECH sampling weights. The reported means for the dependent variables correspond to the pre-reform period (2004-2007).

Table 6: Effect of the health insurance benefit extension on underreporting of salaried earnings, overall and by firm size. Difference in differences estimates.

	(1)	(2)	(3)	(4)
	All registered employees	Small firms (< 10 employees)	Medium firms (10-49 employees)	Large firms (50+ employees)
Panel A. All registered emplo	oyees			
AnyChildren*PostReform	0.0073	0.0342**	-0.0052	-0.0031
	[0.0061]	[0.0152]	[0.0131]	[0.0060]
AnyChildren	-0.0083	-0.0213	-0.0034	-0.0033
	[0.0059]	[0.0138]	[0.0125]	[0.0060]
Observations	50,669	15,151	13,035	22,483
R^2	0.06	0.04	0.03	0.02
Dependent variable mean	0.10	0.18	0.11	0.04
Panel B. Registered employe	es working $\geq 25 h$			
AnyChildren*PostReform	0.0093	0.0409**	-0.0012	-0.0049
	[0.0064]	[0.0166]	[0.0136]	[0.0061]
AnyChildren	-0.0105*	-0.0262*	-0.0042	-0.004
	[0.0062]	[0.0151]	[0.0130]	[0.0062]
Observations	46,222	12,681	12,175	21,366
R^2	0.06	0.04	0.04	0.02
Dependent variable mean	0.10	0.17	0.11	0.04
Panel C. Registered employe	ess working < 25 h			
AnyChildren*PostReform	-0.0128	-0.0051	-0.0799	0.0471
	[0.0224]	[0.0380]	[0.0498]	[0.0287]
AnyChildren	0.0183	0.0143	0.0266	0.0202
	[0.0218]	[0.0346]	[0.0512]	[0.0307]
Observations	4,447	2,470	860	1,117
R^2	0.14	0.15	0.15	0.24
Dependent variable mean	0.15	0.22	0.09	0.03

Notes: The sample includes employed individuals aged 25-55. The data corresponds to the Encuesta Continua de Hogares (ECH) from the 1st semester of 2006 to the 2nd semester 2007 and from the 1st semester of 2009 to the 2nd semester of 2010. The dependent variable is the underreporting of salaried earnings as reported by the individual herself. The controls are the same as those detailed in the notes to Table 2. The regressions are estimated as linear probability models. Regressions are weighted by ECH sampling weights. The reported means for the dependent variables correspond to the pre-reform period (2004-2007).

	(1)	(2)	(3)	(4)	(5)	(6)		
	Benefit-eligible employment	Not working	Unregistered employment	Unregistered Registered employment employment		Underreporting of earnings (small firms)		
Panel A. With time-coho	rt effects							
AnyChildren*PostReform	0.0144*	-0.0175***	0.0022	0.0153**	0.0085	0.0342**		
	[0.0077]	[0.0065]	[0.0060]	[0.0075]	[0.0054]	[0.0152]		
Observations	97,552	97,552	97,552	97,552	56,707	15,151		
Panel B. With AnyChildr	en/covariates int	teractions						
AnyChildren*PostReform	0.0132*	-0.0122*	-0.0004	0.0126*	0.0111**	0.0332**		
	[0.0075]	[0.0063]	[0.0058]	[0.0073]	[0.0053]	[0.0153]		
Observations	97,552	97,552	97,552	97,552	56,707	15,151		
Panel C. With PostRefor	m/covariates inte	eractions						
AnyChildren*PostReform	0.0173**	-0.0165**	-0.0017	0.0182**	0.0072	0.0398**		
	[0.0081]	[0.0070]	[0.0063]	[0.0079]	[0.0059]	[0.0165]		
Observations	97,552	97,552	97,552	97,552	56,707	15,151		
Panel D. Individuals with unchanged payroll tax								
AnyChildren*PostReform	0.0168*	-0.0118	-0.0039	0.0157	0.0107	0.0915**		
	[0.0100]	[0.0143]	[0.0119]	[0.0116]	[0.0272]	[0.0403]		
Observations	40,803	40,803	40,803	40,803	7,023	3,127		

Table 7: Effect of the health insurance benefit extension on labor market outcomes. Robustness and specification checks.

Notes: The sample includes individuals aged 25-55 (all individuals for columns 1-4, conditional on registered employment for columns 5 and 6). The data corresponds to the Encuesta Continua de Hogares (ECH) for the 2nd semester of 2004 to the 2nd semester of 2007 and from the 1st semester of 2009 to the 2nd semester of 2010. For the regressions in Panel A, the data corresponds to the period that spans from the 2nd semester of 2004 to the 2nd semester of 2007. The data for the regressions in column 6 starts in the 1st semester of 2006. "Benefit-eligible employment" refers to registered employment with 25 hours or more per week. The dependent variables and controls are the same as those detailed in the notes to Table 2. The regressions are estimated as linear probability models. Panel A reports results with a full set of birth cohort-semester interactions. Panel B reports results with the AnyChildren indicator interacted with demographic controls. Panel C reports the results from a specification with the PostReform indicator interacted with demographic controls. Panel D reports results by restricting the sample to individuals with salaried (nominal) earnings below 2.5 BPC (the regression sample includes those with zero earnings). The reported means for the dependent variables correspond to the pre-reform period (2004-2007).

I. Tax Revenue	Payroll tax (2007)	Change in payroll tax	Individuals (2007)	Change in number of individuals	Monthly avg. salary (2007 UYU)	Total annual change (million UYU)
A - Mechanical effect						
A. 1 - Childless individuals		0.015	67,122		12,889	155.7
A. 2 - Parents		0.03	159,250		12,689	727.5
B - Behavioral effect						
B. 1 - Parents	0.08	0.03		5,574	12,689	93.4
B. 2 - Underreporting (small firms)	0.08	0.03	7,657	332	8,231	-2.3
C - Total (A.1 + A.2 + B.1 + B.2)						974
			Individuals	Change in	Expenditure	Total annual
			(2007)	number of	per individual	change
II. Expenditure			(2007)	individuals	(2007 UYU)	(million UYU)
D - Mechanical effect						
D.1 - Children acquiring the benefit			242,072		603	1,750.7
D.2 - Reduced public sector exp. on children			162,632		-350	-683.1
E - Behavioral effect						
E. 1 - Parents entering benefit-eligible empl.				5,574	809	54.1
E. 2 - Children of parents entering empl.				8,473	603	61.3
E. 3 - Parents+children leaving the public sector				10,232	-350	-43.0
G - Total (D.1+D.2+E.1+E.2+E.3)						1,140.1
Budget defict, mechanical (D1+D2-A1-A2)						184.5
Net behavioral effect (E1+E2+E3-B1-B2)						-18.7
Total effect (G-C)						165.8
Budget defit/Government expenditures on health	insurance (2	2007)				0.284%
Budget defit/GDP (2007)						0.030%

Table 8: Net impact of the health insurance benefit extension on the government budget: Mechanical effects and behavioral responses.

Notes: The mechanical effects correspond to the changes in revenue due to the changes in payroll tax contribution rates after the reform and to changes in expenditure due to the payment of health insurance premiums and the fall in the use of the publichealth system by those who enrolled in the new benefit after the reform. The behavioral effects take into account the impact of the reform on labor market and tax evasion outcomes. See Section 7 for more details.